

## Do cultural differences affect the quality of financial reporting in the EU? An analysis of Western EU countries vis a vis Eastern EU countries.

Journal:	Journal of Accounting & Organizational Change
Manuscript ID	JAOC-09-2022-0129.R2
Manuscript Type:	Research Paper
Keywords:	Culture;, Financial Reporting;, Classification Shifting;, Eastern European Countries;, Real Earnings Management.

SCHOLARONE"
Manuscripts

Do cultural differences affect the quality of financial reporting in the EU? An analysis of

Western EU countries vis a vis Eastern EU countries.

# March 2023

## ABSTRACT

# Purpose

This study examines the impact of national culture on classification shifting in Eastern European Member States of EU (EEU) vis a vis the Western Member States of EU (WEU). The EEU provides a unique sample to study the quality of financial reporting that we measure with classification shifting given that for more than five decades they were following the model of a centrally planned economy, where market-based financial reporting was absent. Yet, the EEU transitioned to a market-based economy and completed its accession to the EU.

### • Design/methodology/approach

We employ a panel data set of firm year observations from 1996 and 2020 that covers the full transition of EEU. Our empirical analysis is based on fixed effects panel regression analysis where we report a plethora of identifications.

#### • Findings

Our study finds classification shifting in the EEU countries since their transition to the market-based economy, though they have no long record of market-based financial reporting. Our study also notices that cultural factors are associated with classification shifting across all Member States of the EU. We further examine the impact of interactions between cultural characteristics and special items and reveal variability between WEU and EEU. As part of the robustness analysis, we also test the impact of culture on real earnings management measures for both WEU vs EEU confirming the variability of the impact of culture on earnings management.

#### • Originality

This is the first study, to the best of our knowledge, that sheds light on the impact of national culture on classification shifting in EEU of EU vis a vis the "old" WEU of EU.

#### Research limitations/implications

Future research could explore the role of religion differences in WEU vis a vis EEU states as they are also subject to cultural differences.

#### • Practical implications

The findings are important for regulators, external monitors, and investors, as they show that cultural factors affect earnings management with some variability across countries in the EU, and they should be acknowledged in policy making.

Keywords: Culture; Classification Shifting; Eastern European Countries; Financial Reporting;

Real Earnings Management.

JEL: G3, M41, O52, P2

#### **1. INTRODUCTION**

Over the past decades, the process of integration pursued by the European Union (EU) aimed to create a unique economic and political union of western and eastern countries. A wider literature on this process has investigated different aspects of the alignment of economic and political development and democratic approach among different European countries (Persson & Tabellini, 2009; Schmidt, 2019). The dichotomy of Western European Countries (WEU) and Eastern European countries (EEU) has open rooms for management and accounting research on the implication of EEU countries institutional factors change and adaptation to WEU countries in management and accounting practices (Acemoglu et al., 2005; Albu et al., 2020, Mueller & Peev, 2007). The deepening and widening integration process has been pushed by the promotion of certain values among the European Countries with the acknowledgment of the historical differences between WEU and EEU countries. Researchers consider cultural values to be relatively stable societal characteristics (e.g., Inglehart & Baker, 2000; Schwartz, 2006). However, recent research has ascertained that under the considerable amounts of EU reforms initial cultural distance between the founders' average values and the countries acquiring membership has decreased (Akaliyski, 2019).

The influence of how cultural differences in societal groups and values affect the managers' behaviours in manipulating the accounting information (Boahen and Mamatzakis, 2020; Desender et al., 2011; Han et al., 2010; Nguyen et al., 2021) has been widely investigated, using cross countries analysis, including the EU. However, to the best of our knowledge, we find no evidence of research on the different cultural features and its evolution between WEU and the EEU countries on the management attitude to earning management.

This paper provides new international evidence on the impact of national culture on the classification shifting in the EU. Specifically, our research examines whether there are differences in classification shifting between the WEU and the EEU.

The use of classification shifting in our research is justified by its specific characteristics. Classification shifting is distinct from abnormal earnings management and real earnings management (REM) (McVay, 2006). This form of earnings management involves the shifting of individual components of the income statement that are intended to be useful to the investors. The result of this manipulation will not change the GAAP earnings and reduce future (or past) period earnings, as it happens with discretionary accruals and REM (Cohen & Zarowin, 2010; Jarvinen & Myllymaki, 2016; Kothari et al., 2016; Roychowdhury, 2006). Indeed, classification shifting does not involve actions for providing accruals from other periods or selling assets near the end of the year. This is different from abnormal accrual and REM methods. Abnormal accrual consists in the manipulation of earnings management through estimation and accounting methods that have no direct impact on cash flow while real earnings management is to manipulate earnings through operational activities that directly affect cash flow (Sun & Lan, 2014).

Leaving unchanged the GAAP net income, the classification shifting is less detectable by auditors and regulators for the subjectivity of the accounting assumptions applied for the classification of the items (McVay, 2006), making it a desirable tool for engaging in earnings management (Nelson et al., 2002). Additionally, accounting research showed classification shifting has become more pervasive in the post-IFRS era because of less strict regulations (Zalata & Roberts, 2017). The introduction of the IFRS in the EU and the reforms related with a high-quality investor protection environment (audit standards) have aligned the European countries and open more opportunities to manipulate earnings with classification shifting. In this context of regulatory alignment and higher opportunities of core earnings manipulations, it appears relevant to investigate whether the countries cultural differences play a role on the Sonor management decision to apply classifications shifting.

Our empirical examination is focused on the role of culture on classification shifting in EEU, where the culture has been influenced by decades of centrally planned economy and communist values, vis a vis WEU.

The context of the analysis is of a particular interest from the cultural point of view. The EEU have had significant progress towards their convergence to the EU<sup>1</sup>, following a timely and laborious process of reforms that included also specific requirements for their underlying national statistical and financial reporting. This process is still an on-going and subject to certain challenges, mostly related to institutional heterogeneity between the WEU and EEU (Albu et al., 2020) which is interrelated with the differences in cultural values (Albu et al., 2020; Eder et al., 2017; Nguyen et al., 2021). This paper follows from previous studies regarding the importance of culture for EEU (Albu et al., 2020; Eder et al., 2017) and proposes to examine whether they explain earnings management in the form of classification shifting and real earnings management as a robustness analysis.

Our research builds a panel of cross-country data set with both culture and earnings management variables. We follow the seminal study of Han et al. (2010) that investigated how managers' earnings discretions relates to their value system (i.e., culture). Han et al. (2010) applied Gray's (1988) model of accounting values based on Hofstede's (1980) societal values, which are identified in individualism, uncertainty avoidance, power distance, masculinity. They analysed the difference between the U.S and non-U.S. companies and their findings showed that uncertainty avoidance and individualism dimensions of national culture are, respectively, negatively, and positively associated with earnings manipulation' behaviours. They extended Guan et al. (2005), which used firm-level data in 5 Asian-Pacific countries to evaluate the association between Hofstede's (1980) cultural dimensions' variables and discretionary

<sup>&</sup>lt;sup>1</sup> They overcame their recession in the early 1990's, created by their political and economic transformation after the fall of the Berlin Wall, that paved the way to market-based economy and to their accession to the EU.

accruals. Following the same framework, the main literature on the influence of cultural dimensions on earnings management manipulation reports conflicting results (Kim et al., 2017; Paredes & Wheatley, 2017). Interestingly, concerning the WEU and EEU countries, no such study exists in the literature. Furthermore, the literature on earnings management and cultural dimensions is very limited (i.e., Desender et al., 2011) and without due cognisance of classification shifting.

Although prior cross-countries studies have demonstrated the differences in engaging in earnings management, accrual-based and REM (i.e., Leuz et al., 2003), only a few of them have analysed the extent of classification shifting (Behn et al., 2013; Haw et al., 2011). There are some cross-countries studies that research classification shifting with institutional features with Behn et al. (2013) showing that classification shifting is negatively affected by string investor protection, and Haw et al. (2011) revealed that classification shifting behaviour is associated with code law countries. However, we do not have knowledge of classification shifting investigated at the EU level and within it.

In order to explore the causal effects of countries' cultural dimensions on classification shifting, we take advantage of the fact that the EEU form a unique sample of countries that shared distinct culture values that depart from WEU. Most countries of the EEU are former members of the East European block and individualism was not a shared cultural value. Clearly, after historical events such as the collapse of Berlin wall and what followed, those countries switched to pro-western cultural values while they adopted reforms in financial reporting, legal institutions, and corporate governance. Despite those reforms, it is of interest to investigate whether there is variability in certain cultural characteristics in EEU vis a vis WEU. In addition, the EEU show a mix of historical and cultural influences with the presence of Western individualistic philosophies, geographical and political factors that have encouraged collectivism (Varnum et al., 2008). Therefore, the EEU provide a most noteworthy contrast to

WEU. The efforts in transforming East Europe's governance and institutional practices have created a heterogeneous picture in recent times. These institutional changes have inevitably altered the countries' cultural dimensions; although these aspects are certainly apt for reducing the previous cultural approach produced by the paternalistic character of the regime, it will take time to align these countries to the WEU's culture of open markets. Moreover, these countries' reforms and development are regularly instrumentalized for political purposes or are otherwise subject to political influence in the way they are implemented (Haselmann et al., 2010). Therefore, we test how the cultural dimensions are transmitted to the private sector and affect managers' behaviour.

Our study applies Hofstede's model (1980, 2001) of the cultural dimensions to explain the implication of individualism, uncertainty avoidance, masculinity, and power distance on the classification shifting in the EU's market. Cross cultural research has been deeply informed using Hofstede model because of its "clarity, parsimony, and resonance with managers" (Kirkman et al., 2006, p. 286). However, several studies have questioned the applicability of Hofstede's cultural value scores, pointing out as major criticism the failure of the model in capturing the change of culture over time (Kirkman et al., 2006; Tang & Koveos, 2008). Inspired by Hofstede et al. (2010) and according to Beugelsdjik and Welzel (2018)'s approach, we use World Value Survey (WVS) 's waves data to capture the cultural dimensions of the EU countries. Furthermore, our study introduces Hofstede's "long-term orientation" dimension (Hofstede, 2010), which has been neglected in the previous analyses of the effects of cultural dimensions and earnings manipulation.

The data employed for the measurements of the classification shifting and other proxies of earnings management have been collected from Compustat Global Database. The data set includes 79,570 firm-year observations for the period from 1996 to 2020 across all European

 countries (EU28). Our analysis is in line with prior studies (Behn et al., 2013), with a minimum of 10 firm-year observations used to estimate abnormal core earnings.

In terms of results, our study highlights a positive relationship between classification shifting and special items confirming the existence of misclassification both in WEU and EEU. To this end, it appears that the EEU countries have been catching up in terms of earnings management behaviours using the classification shifting.

When considering cultural factors, it appears evident how some cultural differences are still alive and the two blocs that are part of the EU, EEU and WEU, and they affect the manager's manipulation behaviours. Although masculinity and power distance showed the same impact on both the EU blocs analysed, due to the strict regulatory environment of the market and the application of IFRS, the cultural dimensions of uncertainty avoidance and independence differentiate the two blocs in their approach to the earnings management. The different level of acceptance of the uncertainty avoidance in the EEU, driven by years of regime and higher level of hierarchy with a strong control of all the aspect of social life, has a negative relation with the classification shifting (positive for the WEU), showing how the fear of the uncertainty act as a deterrent to manipulate the accounting figures. The same result has been found in relation with the independence where the relationship is negative with the earnings manipulation in EEU. This is the mirror of the lasting presence of the culture of "obshchina" (meaning community that entails solidarity across all community members, Kuchma, 2003), which enhances the awareness of the manager on the damage that earnings manipulation could bring on the firm, the efficiency of the market and the society at large.

We provide also additional evidence expanding earnings management measurement to include REM.

The paper proceeds as follows. Section 2 discuss the main studies on the earnings management, classification shifting and cultural values with the developed hypothesis

contextualized for EEU; section 3 describes data, sample selection, and descriptive statistics and section 4 provides research design and discusses empirical methodology. Section 5 discusses empirical results. Section 6 presents robustness checks and section 7 offers the conclusion.

#### 2. Literature Review and Hypotheses

#### 2.1 Earnings management and culture values

Leuz et al. (2003) established the existence of the negative relationship between earnings management and countries' institutional factors, with the level of investor rights, based on firm-level data in 31 countries. Fernandes and Ferreira (2007) repeated the study by Leuz et al. (2003) with an analysis based on firm-level data in 47 countries, and their findings affirm that the accruals-based earnings management in a country is influenced by specific firm-level characteristics. Their research noted that external financing and firm valuation are negatively related to proxies for accruals-based earnings management.

The significance of the country's specific factors in the field of research on earnings management opened room for research on the effects of national culture on the company's behaviour and capital markets. Several studies have noted the implication of countries' cultural features on the financing choices (Fan et al., 2008), on the merger and acquisition (Ahern et al., 2015), and on corporate risk-taking approach (Li et al., 2013). These findings, confirm the interrelationship between a country's institutional factors and culture (Hutchings & Michailova, 2006) with people's behaviours shaped by their national cultural dimensions. Accordingly, managers' attitude toward specific practices "reflects the cultural values of the society in which they live and work" (Paredes & Wheatley, 2017, p. 43). Managers' opportunistic behaviours are reflected in the use of different earnings manipulation activities in financial reporting (Kim & Park, 2014). Most of the accounting literature on interrelationship between earnings

management and national culture has focused on two main categories of accrual manipulation: discretionary accruals and REM (Healy & Wahlen, 1999). The discretionary accruals occur when managers manipulate the company's earnings by using the accounting discretion allowed under accounting standards (Jackson, 2018). The REM involves decisions on the timing and the structure of the earnings-related to actual business activities, such as R&D or advertising expense, aiming to meet current-period financial reporting targets (Cohen et al., 2020; Zang, 2012).

The accounting research has widely investigated how the management decision to manipulate earnings is affected by cultural differences (Guan et al., 2005; Leuz et al., 2003). This stream of the literature has rooted on the Hofstede's cultural model which defines culture as "the collective programming of the mind which distinguishes the members of one human group from another" (Hofstede, 1980, p. 25)<sup>2</sup>.

The model developed in 1980 identifies power distance, individualism, uncertainty avoidance, masculinity and long-term orientations as cultural dimensions which characterize the culture of different countries (Hofstede, 1980; Hofstede et al., 2010).<sup>3</sup>

The degree of acceptance of inequality or equality in a country, organization, and institution has been measured by Hofstede with the cultural dimension known as power distance. Individualism versus collectivism is the cultural dimension used by Hofstede to measure the degree to which people are assimilated into societal groups or interpersonal

<sup>&</sup>lt;sup>3</sup> The model was built on the analysis of a survey of employees' attitudes in all subsidiaries of IBM using 117 questionnaires in 40 countries. Subsequently, Hofstede (1991) applied the survey in 10 countries and three regional groupings of countries, not previously included in his cultural model. The new investigation led to the introduction of a new cultural variable, known as long-term orientation, which was identified as specific dimension revealed in the survey of Chinese employees and managers (Hofstede et al., 2010).



<sup>&</sup>lt;sup>2</sup> The cultural dimensions' model provides a quantitative score for each of the cultural variables identified by Hofstede (2010). Based on Hofstede's model, Gray (1988) developed a framework that classifies the different accounting practices based on cultural differences within countries – what he terms "national" accounting systems. This framework, subsequently expanded by Doupnik and Tsakumis (2004), uses Hofstede's (1980) societal values to identify system of accounting values that influence the accounting practices and outcome. Gray's (1988) model has been used to explain how the cultural differences may influence the outcome and the behaviours of accounting players, and their attitude to apply the same set of rules in different ways.

relationships. Hofstede's uncertainty avoidance dimension refers to the country's attitude to tolerate uncertain and ambiguous situations. The cultural dimension of masculinity measures the presence of the traditional role of male in a society. The distribution of power, control, and success between genders and the acceptance of predominance of the male and female role characterized by specific stereotypes will affect the cultural model enforced in a country. Long-time is a universal value that relates to how we see the influence of the past, present and future in our life: how far we plan; how quickly we expect our result and rewards; how important we consider saving and spending, etc. (Hofstede et al., 2010).

In accordance with this framework, Han et al. (2010) investigated the managers' discretional behaviour in manipulating a company's accruals and how it relates to their value system (i.e., culture). They analysed the difference between U.S and non-U.S. companies, and their findings showed that uncertainty avoidance and individualism dimensions of national culture are, respectively, negatively and positively associated with earnings manipulation' behaviours. Using a firm-level data in 5 Asian-Pacific countries, they expand the literature on cultural differences implications on the earnings management, following Guan et al. (2005) attempt of evaluating the relationship between the use of discretionary accruals and the cultural dimensions, as identified in Hofstede's model (1980). Their findings affirm a negative relationship between individualism, uncertainty avoidance, and discretionary accruals. In a related study on 49 countries, Callen et al. (2011) find that individualism is positively related to earnings management and that uncertainty avoidance has a negative relationship with accruals manipulation. Another cross-country analysis (Kim et al., 2017), using a sample of 38 countries, observe that higher levels of uncertainty avoidance, individualism, and power distance are associated with fewer earnings manipulation's behaviours, while higher levels of masculinity lead the managers to engage in more earnings management. Paredes and Wheatley (2017) extended the literature regarding the interrelationships between national culture and

REM for emerging countries, revealing a negative association between REM and Hofstede's (1980) measures of individualism, masculinity, and uncertainty avoidance, but a positive association with power distance. This interest on the implication of Hofstede's cultural dimensions has been concentrated mainly on abnormal accruals and REM behaviour, with only few studies related with classification shifting (Boahen and Mamatzakis 2021). These studies are mainly related with USA and with developed countries where the institutional factors seem to enhance the use of classifications shifting. As for today, there are no studies applying the Hofstede's model to the WEU and EEU to evaluate how their historical differences are still mirrored in the diverse management behaviours related with classifications shifting. This is the aim of this study. Based on the acknowledgement that the EEU integration process has inevitably flattened the difference in the institutional factors among the countries part of the Union, the analysis of the Hofstede's dimensions on the management behaviours appears to be significant for understanding if the integration has happened also at cultural level. Additionally, the use of the classification shifting in this context is justified by the implications of the harmonization process in term of financial statements with the application of IFRS which open different opportunities to manage the core earnings and without changing the final figures of the Income statements as for the accrual and REM (Zalata & Roberts, 2017).

#### 2.2 Cultural values in Eastern European Countries and hypothesis developments

National culture shapes country-wide values, legal systems, and institutions as well as economic resource allocation (Stulz & Williamson, 2003). These factors characterize the investment's environment, which become the base for increasing the level of trust in the capital market (Bhattacharya et al., 2007; Stiglitz, 1989).

The EEU represent on this aspect a unique sample to investigate the unprecedented changes of national culture undertaken during the last two decades. After the fall of the Berlin

wall, the European Union was enlarged to include former East European countries' block; these countries rapidly had to adapt to the European legal framework and had to change their cultural approach, which had been crystallized under the communist system of centrally planned economies. The task of transiting from centrally planned to market-based economy is a herculean one, as the latter was in operation for over five decades. One might suspect that certain cultural characteristics could persist in former communist EEU. However, to this day there is no evidence for such persistence.

Indeed, the changes in terms of the legal environment are evident, considering that the reception of the different EU regulations and the transformation in the cultural approach have required time and they are still not easy to measure or evaluate. EEU experienced half a century of Communist rule, which posits contextual and situational causes for historical events and individual behaviour. It was a repressive regime that constrained individuals' freedom, affecting the level of trust in the institutions, and encouraging collectivist thinking and interdependent behaviour (Varnum et al., 2008).

Additionally, the collapse of communism in EEU produced great social and economic uncertainties for its newly independent states. The life, values, and culture of the population in these countries were affected by the rush to develop "market" economies, privatization, and marketization, which took place far faster than legislative reforms. Moving from a communist to a democratic political system, from a collectivistic to individualistic society, from centrally planned to market-based economy have inevitability changed the previous cultural characteristics of the EEU with a fast catch up with the values and the beliefs of the WEU (Akaliyski, 2019). These changes have naturally influenced the management behaviours and therefore their attitude in manipulating the earnings to meet the investors' expectations Sol Cr (Doupnik, 2008; Leuz et al., 2003).

#### 2.2.1 Power distance

Power distance is a cultural dimension which represents the degree to which less powerful individuals in organizations expect and accept the unequal allocation of power (Halkos & Skouloudis, 2017). The higher is the score that measures the power distance dimension, the higher is the willingness of the members of the organization or institutions to accept an unequal power distribution (Fidrmuc & Jacob, 2010; Li et al., 2013).

In countries with the higher score in terms of power distance, Waldman et al. (2006) noted a social pressure on managers to be less opportunistic, which reduced the likelihood of them engaging in unethical earnings management practices. Hofstede et al. (2010) observed that power distance score tends to be higher for EEU than for WEU. A great role in the power perception of the EEU was played by the democratic centralism approach, investigated by Kets de Vries (2000). The author noted that democratic centralism gave the opportunity to all the members of the community to participate in the discussion related to policies and issues and vote for the leadership. However, after the election of the leader, he centralized all the powers and was given the legitimacy to carry out his/her chosen policies in an autocratic manner without opposition. Other studies have confirmed these results (Bakacsi et al., 2002; Schwartz & Bardi, 2001) with an emphasis on the predominance of hierarchical values in EEU, in contrast with autonomous values. In their analysis of the influence of national cultural dimensions and earnings management behaviours, Kim et al. (2017) find that managers in power distance engage less in earnings management. This discussion leads to the following hypothesis:

**H1:** Higher power distance, as associated with EEU, has a negative impact on classification shifting.

#### 2.2.2 Individualism

In societies where there are loose ties, limited links to the extended family, and where the immediate family and the personal interests are prominent, the score of individualism is high (Davis & Abdurazokzoda, 2016; Hofstede, 1980; Hofstede et al., 2010; Klasing, 2013). The findings of several studies affirm that East European culture is less individualistic than West European (Hofstede, 1980; Kolman et al., 2003). This can be explained by the East European authoritarian practices which structurally imposed the collectivist culture that shifted the authority perceptions away from individualism. The affirmation of the collectivist approach and power perception of the population living in the EEU has been facilitated by the rural value of "obshchina" (community), which enforced the strict control on the community but also the solidarity across community members (Kuchma, 2003). The Eastern European's cultural approach of interdependence, in which members of the community provide support and moral guidance, can be considered the legacy of obshchina (Kets de Vries, 2000). Hofstede et al. (2010) suggested that accounting plays a significant role in conducting business in individualistic countries than social collectivistic countries. Managers are more incline to show optimistic performance to meet analysts' expectation (Han et al., 2010). According to these assumptions we develop the following hypothesis:

H2: Low level of individualism, as associated with EEU countries, has a positive 07. impact on classification shifting.

#### 2.2.3 Uncertainty avoidance

Uncertainty avoidance as a cultural dimension is associated with the degree to which the nation or society is tolerant of uncertainties or ambiguities (Ho et al., 2012; Hofstede, 1994; Ringov & Zollo, 2007). Nations or societies with high scores in this dimension are more inclined to being uncomfortable with new or uncertain situations and use laws, rules, and codes of conduct as a form of protection, aiming to mitigate uncertainties (Hofstede, 1980, 1994).

Contrary to this, nations or societies with low scores in this dimension exhibit more flexibility in their attitudes and behaviours and are more likely to get more involved in uncertain situations (Hofstede, 1980). In their analysis, Hofstede et al. (2010) find a higher score for uncertainty avoidance in EEU rather than in WEU. The finding supports the historical and political development of these countries, considering the strict hierarchy and the rigid regime's control of social life. Consequently, managers in these countries are more risk-averse and careful in their decision to avoid possible and unpredictable outcomes. Therefore, societies with a high level of uncertainty avoidance develop more conservative accounting systems and behaviours (Gray, 1988). The influence of conservatism on the managers' behaviours could either lead to greater use of earnings management to side-step the risk associated with the potentially negative assessment of accruals management, and disincentivise the managers' engagement in earnings manipulation given its possible damaging effect on firm performance in the future. In their analysis Han et al. (2010) and Guan et al. (2005) highlight a negative relation between discretionary accruals and uncertainty avoidance. This discussion leads to the following hypothesis:

H3: Uncertainty avoidance, as associated with EEU, has a negative impact on 2.12 classification shifting.

#### 2.2.4 Masculinity

Nations or societies assumed as masculine are characterised by competitiveness, assertiveness, and individual achievement, with power and success envisaged as material returns (Ho et al., 2012; Hofstede, 1980). Under this cultural dimension, winning and being the best are the goals.

Thereby, higher score in terms of masculinity in a country are generally linked with a management cultural approach driven by ego-orientation, valuing money, and property values, which prioritizes high economic growth and prefers high pay. In Hofstede et al. (2010),

masculinity is higher for EEU than WEU. Kim et al. (2017) findings support this discussion with evidence of the higher influence of the level of masculinity on the managers' earnings manipulation behaviours. Following the above discussions, we state the following hypothesis to be tested:

H4: Masculinity, as associated with EEU, has a positive impact on classification

shifting.

### 2.2.5 Long-Term Orientation

Human societies have always been organized within space and time. The former is related to the relationship between humans and nature. The latter focuses on how we perceive and vision our life activities across the time spectrum. At the collective level, time spectrum exerts different degrees of influence in various societies, creating two orientations on this value-dimension: short- and long-term time orientation. In a nutshell, a focus on the past and on the present would lead more towards short-term time orientation, and a focus on the future will lead more towards long-term time orientation.

This dimension symbolises a nation that is more oriented towards the future and that attaches importance to the economy and persistence (Halkos & Skouloudis, 2017; Hofstede et al., 2010).

Contrarily, a short-term oriented nation appreciates tradition and abides by its social obligations (Hofstede, 1994) with focus on immediate outcomes, that may include satisfaction and happiness in the present (Thanetsunthorn & Wuthisatian, 2018). Hence, we formulate the following hypothesis:

**H5:** The degree of long-term orientation in EEU will assert a positive impact on classification shifting.

#### 

#### 3. Data and Descriptive Statistics

# 3.1. Measuring Cultural Dimension Score – The World Value Survey and Hofstede's model

We collect national dimensions of culture datasets from the WVS according to Buegelsdjik and Welzel (2018)'s approach. The WVS is a common questionnaire developed to study and understand the values across cultures and across time (seven waves from 1981 to 2021). The data are collected from 1981 until the present over 100 countries on all six continents and are focused on beliefs, values, economic development, democratisation, religion, gender equality, social capital, and subjective well-being. Therefore, the WVS provides data on socio-cultural and political change worldwide, and it has been widely used in the cross-countries cultural analysis in different disciplines and in accounting, including Brochet et al. (2019) and Knechel et al. (2020).

Hofstede et al. (2010) have developed one of the cultural dimensions (long-term orientation) using some of the questions in the survey. Of relevance to this paper are people's responses to the questions on the ease with which they can be correlated with the Hofstede's cultural dimensions as for Beugelsdjik and Welzel (2018). More specifically, we use five WVS questions which capture power distance scores (PD), individualism scores (IND), uncertainty avoidance scores (UAI), masculinity scores (MASC) and long-term orientation scores (LTO) as for the table 1.

#### [Table 1 near here]

#### 3.2. Data Collection for Classification shifting

All financial data comes from Compustat. The full sample consists of 79,570 firm-year observations for the period from 1996 to 2020 across EU28 countries. In some detail, our sample includes what we call the WEU Member states which are: Belgium, France, Germany,

Italy, Luxembourg, Netherlands, Denmark, Ireland, Greece, Portugal, Spain, Austria, Finland, Sweden. We also include in WEU the UK despite the recent developments due to Brexit, and we also include Cyprus and Malta, though there are not complete data sets for these two countries. The EEU are Czechia, Estonia, Hungary, Latvia, Lithuania, Poland, Slovakia, Slovenia, Bulgaria, Romania, and Croatia. It should be noted that our sample is unbalanced as the various EEU and WEU Member states enter the EU at different years.

In line with prior studies (Behn et al., 2013), countries require a minimum of 10 firmyear observations to qualify for inclusion in the sample for this study. Again, all firm-years with missing observations for all variables were deleted. Furthermore, to effectively use sales as a deflator for most of the variables and avoid the creation of outliers, we exclude any firm-year observation with sales revenue less than \$500,000 in line with prior studies (Fan et al., 2010; Haw et al., 2011; McVay, 2006;). Consistent with prior studies (Behn et al., 2013; Haw et al. 2011) in classification shifting, we use Fama and French (1997) four-digit (SIC) Industry Classification codes.

#### 3.3 Firm specific data

Our main variable is classification shifting ( $UE\_CE_{i,t}$ ), a widespread malpractice that inflates core earnings by altering the presentation of income statement line items without affecting bottom-line income. For classification shifting, we have relied on previous literature (such as McVay, 2006). and for REM ( $REM1_{i,t}$  and  $REM2_{i,t}$ ), we have based our variables on Cohen et al. (2020), but we will provide detail explanation of these variables in paragraph 4.

We consider several firm specific control variables. Firstly, we consider the presence of income-decreasing special items (*SPITEM*<sub>*i*,*i*</sub>) (Boahen & Mamatzakis, 2020). Ashbaugh et al. (2003) and Callen et al. (2011) show that size (*SIZE*<sub>*i*,*i*</sub>) of firm matters, as large firms are less likely to manipulate reported core earnings compared to small firms. Book to market value

( $MBV_{i,t}$ ) controls for the possible impact of market capitalisation. Zang (2012) shows that firms manipulate reported earnings to improve financial management. To capture such effects, we opt for leverage ( $LEV_{i,t}$ ). Similarly, performance could be of importance, and we therefore include profitability ( $ROA_{i,t}$ ) (see Kothari et al., 2005; Zalata & Roberts, 2015). Zalata and Roberts (2015) show that firms with low performance could be more likely to apply classification shifting. Athanasakou et al. (2009) and Doyle et al. (2003) show that cash flow is affected by growth of firms, thus we opt for firm growth. Myers et al. (2007) document that the firms that had preceding positive earnings are more likely to manipulate earnings to keep the consecutive earnings growth trend, and for this reason we control for economic development ( $GROWTH_{i,t}$ ). Lastly, we include total common equity divided by equity shares, that is  $EXPOSURE_{i,t}$ , because prior evidence (Barton & Simko 2002; Cheng & Warfield 2005) shows that large amount of outstanding shares would indicate small earnings per share and thereby more incline to classification shifting.

### 4. Research design and empirical methodology

#### 4.1. Measuring Classification Shifting

In line with prior studies (Behn et al., 2013; Boahen, 2017; Boahen & Mamatzakis, 2021; Fan et al., 2010; Haw et al., 2011; McVay, 2006), we model the core earnings ( $CE_{i,t}$ ) at normal level for firm *i* and year *t* as:

$$CE_{i,t} = \beta_0 + \beta_1 CE_{i,t-1} + \beta_2 ATO_{i,t} + \beta_3 ACCRUALS_{i,t-1} + \beta_4 \Delta SALES_{i,t} + \beta_5 NEG_{\Delta}SALES_{i,t} + \varepsilon_{i,t}$$
(1)

where core earnings with lag is  $CE_{i,t-1}$ ; asset turnover is  $ATO_{i,t}$  and measures sales divided by average net operating assets; accruals with lag is  $ACCRUALS_{i,t-1}$ ; change in sales

with lag is  $\Delta SALES_{i,i}$ ,  $NEG_{\Delta}SALES_{i,t}$  is an index that takes value of one if change in sales is less than zero, and zero otherwise.

Given the criticism of Fan et al. (2010) that report issues with endogeneity with the current level of some of the variables, the above model does not use current accruals but its lagged value on the right-hand side as in McVay (2006).<sup>4</sup> Equal wise, given that core earnings are reported (see Fan et al., 2010) not to affect current level of normal core earnings, we opt for their lagged values. Nissim and Penman's (2001) study shows that asset turnover ( $ATO_{i,t}$ ) affects core earnings, thus we include it in equation 1. Therefore, we include the change in sales  $\Delta SALES_{i,t}$  and negative change in sales  $NEG_{\Delta}SALES_{i,t}$  as in McVay (2006) model.

We employ the equation 1 using regression analysis and estimate the parameter for each firm in our sample, so as to measure the unexpected core earnings  $(UE\_CE_{i,t})$  which are the reported core earnings  $(REP\_CE_{i,t})$  minus normal core earnings  $(CE_{i,t})$ . The unexpected core earnings  $(UE\_CE_{i,t})$  is the main variable that we use to test whether there is classification shifting.

Following from Ashbaugh et al. (2003), Behn et al. (2013), and Fan et al. (2010), we then test for the existence of classification shifting by estimating the following equation 2:

 $UE\_CE_{i,t} = \beta_0 + \beta_1 SPITEM_{i,t} + \beta_2 SIZE_{i,t} + \beta_3 BMV_{i,t} + \beta_4 LEV_{i,t} + \beta_5 ROA_{i,t} + \beta_6 GROWTH_{i,t} + \beta_7 EXPOSURE_{i,t} + Year and Country Dummies + \varepsilon_{i,t}$ (2)

where unexpected core earnings is  $UE\_CE_{i,t}$ ; income-decreasing special items multiplied by minus one is  $SPITEM_{i,t}$ ; firm size is  $SIZE_{i,t}$ ; book to market value is  $BMV_{i,t}$ ;

<sup>&</sup>lt;sup>4</sup> Note that in line with Behn et al. (2013) and Francis and Wang (2008), accruals equal earnings before extraordinary items minus operating cash flows divided by total assets with one lag. Earnings before extraordinary items equal net income minus extraordinary items. Also note that operating cash flows equals the sum of earnings before extraordinary items, depreciation, and amortization, change in deferred income tax, change in untaxed reserve, change in other liabilities, minority interest minus current accruals.



 leverage  $LEV_{i,t}$ ; return on assets is  $ROA_{i,t}$ ; growth counts for  $GROWTH_{i,t}$  and total common equity divided by equity shares  $EXPOSURE_{i,t}$ .

It is worth noting that classification shifting exists when special items would positively affect unexpected core earnings. Therefore, the main parameter estimates of equation 2 concern  $\beta_1$ . If  $\beta_1$  is positive and statistically significant, it would imply that there is classification shifting, which is misclassification of core expenses into special items.

#### 4.2. Real Earnings Management

We also consider in our identification the impact of culture on REM. Regarding real activities earnings management, we opt for two measures. The first measure for real activities management ( $REMI_{i,t}$ ) is estimated using the abnormal cash flows ( $ABNOR\_CASH_{i,t}$ ), the abnormal discretionary expenses ( $ABNOR\_DEXP_{i,t}$ ) and the abnormal production costs ( $ABNOR\_PCOST_{i,t}$ ) for each firm and industry classified by its two-digit SIC code (see also Dechow et al., 1996). The  $REMI_{i,t}$  is estimated as the sum of abnormal discretionary expenses ( $ABNOR\_DEXP_{i,t}$ ) multiplied by minus one and abnormal production costs ( $ABNOR\_DEXP_{i,t}$ ). Note that high values of  $REMI_{i,t}$  would imply evidence that firms engage in earnings management and thus underreport expenses to boost earnings.

The first step to estimate  $REM1_{i,t}$  is to derive the abnormal cash flows  $(ABNOR\_CASH_{i,t})$  which is provided, similarly to classification shifting methodology, as deviations from predicted values from the industry-year regression. More in detail, we use industry-year regression to estimate cash flows over lagged total assets as follows:

$$\frac{\text{CASFO}_{i,t}}{\text{TA}_{i,t-1}} = \beta_0 + \beta_1 \left( \frac{\text{SALES}_{i,t}}{\text{TA}_{i,t-1}} \right) + \beta_2 \left( \frac{\Delta \text{SALES}_{i,t}}{\text{TA}_{i,t-1}} \right) + \varepsilon_{i,t}$$
(3)

where  $CASFO_{i,t}$  is the cash flow from operational activities.  $SALES_{i,t}$  represents annual sales revenue and TA<sub>i,t</sub> total assets is the aggregate of both non-current and current assets, whilst  $\Delta SALES_{i,t}$  is change in sales. The figure for  $(ABNOR\_CASH_{i,t})$  is multiplied by minus one, in line with previous studies (Roychowdhury, 2006; Zang, 2012).

The abnormal production costs (ABNOR  $PCOST_{i,t}$ ) are derived as deviations from predicted values from the industry-year regression. Following Cohen and Zarowin (2010) we estimate abnormal production costs using the following equation:

$$\frac{PCOST_{i,t}}{TA_{i,t-1}} = \beta_0 + \beta_1 \left(\frac{SALES_{i,t-1}}{TA_{i,t-1}}\right) + \beta_2 \left(\frac{\Delta SALES_{i,t}}{TA_{i,t-1}}\right) + \left(\frac{\Delta SALES_{i,t-1}}{TA_{i,t-1}}\right) + \varepsilon_{i,t} \quad (4)$$

where  $PCOST_{i,t}$  is the aggregate of cost of sales and change in inventory during the year.

The normal level of discretionary expenses as in Cohen and Zarowin (2010) and Roychowdhury (2006) is derived from the following equation:

$$\frac{D_{EXP_{i,t}}}{TA_{i,t-1}} = \beta_0 + \beta_1 \left(\frac{SALES_{i,t-1}}{TA_{i,t-1}}\right) + \varepsilon_{i,t}$$

where  $DEXP_{i,t}$  is the sum of advertising expenses, R&D expenses, and general and administration expenses<sup>5</sup>;  $\frac{SALES_{i,t-1}}{TA_{i,t-1}}$  is annual sales revenues over lagged total assets.

<sup>&</sup>lt;sup>5</sup> As seen in Cohen and Zarowin (2010) and Zang (2012), we give the value of zero if selling and general expenses are available, but advertising and R&D expenses are not.

It is worth noting that we opt for lagged sales to avoid issues with endogeneity. Like in the above models, abnormal discretionary expenses ( $ABNOR\_DEXP_{i,t}$ ) are estimated using the parameter estimates of equation (5) from industry-year regression.

For robustness analysis in terms of dealing with possible model misspecification due to measurement errors, we also employ a second measure of REM,  $REM2_{i,t}$ . This time, to measure  $REM2_{i,t}$  we employ abnormal cash flows  $(ABNOR\_CASH_{i,t})$  times minus one and abnormal discretionary expenses  $(ABNOR\_DEXP_{i,t})$  times minus one. As before, high values of  $REM2_{i,t}$  would indicate earnings management.<sup>6</sup>

#### 5. Empirical Regression Results and Discussions

#### 5.1. Main Descriptive Statistics

Table 2 reports the main descriptive statistics for *UE\_CE*, *REM1* and *REM2*, as well as the main variables of our identification for testing our hypotheses of earnings management for both WEU countries and EEU.

# [Table 2 near here]

#### 5.2. Testing Existence of Classification Shifting

Because our focus is on examining the impact of the time-varying process of transitioning firms from EEU of the EU from central planning economy to market-based economy on classification shifting, we choose fixed effects panel regression analysis as this modelling is unbiased. To achieve this, our fixed effect model includes the unique characteristics of each firm, per Member State, that may affect classification shifting. Instead, our results would have been

<sup>&</sup>lt;sup>6</sup> Cohen and Zarowin (2010) and Haga et al. (2018) show that both measures might suffer from aggregation bias. Therefore, some caution is warranted.

biased and of poor significance if we had used a simple panel regression without fixed effects. We can manage firm and country level heterogeneity using fixed effects. In essence, we can observe the impact of the transition to market-based economy on classification shifting for EEU thanks to our fixed effect panel model. This impact is net of time-invariant characteristics that may result in biased estimations. Because each firm in our sample is different and unique, our fixed effects model assumes that the error term for each firm and the constant term that regulates firm-specific attributes should be uncorrelated with each other.

Initially, we test UE\_CE with SPITEM to check for the existence of classification shifting in the full sample (EU28), and then in the 2 subsamples (WEU and EEU) as reported in Table 3. This is preliminary results to establish whether there is classification shifting in EU. Fixed effects panel regressions are estimated to control for heterogeneity across firms. Note that we also opt for robust standard errors with 50 bootstrap replications to control for model misspecification. In the remaining empirical modelling, we augment the model to include additional variables that affect classification shifting.

It is worth noting that the main variable to identify the existence of classification shifting is the income-decreasing special items multiplied by minus one (see  $SPITEM_{i,t}$ ). In general, a positive parameter estimation for  $SPITEM_{i,t}$  would indicate the existence of classification shifting (Ashbaugh et al., 2003; Behn et al., 2013; Fan et al., 2010).

The coefficient of SPITEM is positive (for EU28 0.205, for WEU 0.287 and for EEU 0.476) and significant at 1% level, indicating that some firms in Europe and in the two sub samples inflate core earnings by misclassifying core expenses into special items. It is striking that we find evidence of classification shifting for the EEU states (also for the old ones) as the parameter estimate of SPITEM is positively and statistically significant at 1% level, while also significant in magnitude. This is the first time, as far as we are aware, that evidence of classification shifting is reported for the EEU.

These findings suggest that when income is moved upward or core expenses are moved downwards on the income statement, unexpected core earnings increase (UE\_CE), suggesting that firms do not only report true and fair performance but are also embarked on misclassification of special items to increase reported core earnings. Our findings are in line with previous literature indicating that unexpected core earnings increase with special items (Fan et al., 2010; McVay, 2006) mainly focused on Western countries. However, the result included in column 3 in Table 3 shows that classification shifting is present in the EEU and, that the magnitude of SPITEM coefficient is larger compared to WEU. The EEU, though they did transit from centrally planned economy to market based economy just two decades ago, appear to have quickly adopted earnings management practices of their colleagues in the WEU. Of course, this might not come as a surprise, given that the EEU have been overzealous in their convergence to the EU and the anger in terms of other characteristics, such as orientation to profit maximizations supported by opportunities raised by an open capital market.

### [Table 3 near here]

#### 5.3. Testing Relationship between Culture Variables and Classification Shifting

To test our hypotheses regarding the impact of culture, we run panel regressions that reveal the impact of underlying components of culture on UE\_CE. In each model we include control variables to assist identification.

Table 4 reports a positive relationship between *UE\_CE* and uncertainty avoidance (*UAI*) for the EU28 and for WEU, while the result is negative and significant for EEU. The results for the EU28 and WEU are in line with the literature (Han et al., 2010; Kim et al., 2017).

Our findings are in line with Hypothesis 3 that foresees a negative association between uncertainty avoidance and classification shifting. For EEU the negative impact of *UAI* on

*UE\_CE* could be explained by the persistence of centrally planned economy culture whereby laws and regulations were very restrictive and there was low uncertainty and deviations from the norm that gave fewer opportunities for opportunistic behaviour (see also Han et al., 2010 and Guan et al., 2005). Additionally, the culture of "obshchina" might enhance the awareness of the manager on the damage due to earnings manipulation on the community. On the other side, the positive relationship between UAI and WEU, affirms the prior reported effect that managers are usually more prone to managing earnings to meet market performance and persuade their own personal goals when facing uncertainty (Bermpei et al., 2022; Guo & Jiang, 2011; Hirshleifer et al., 2009).

*ROA* and *LEV* have positive and significant values in all models. These results are in line with previous literature (Zalata & Roberts, 2015), which indicate that firms with strong performance could be engaged with classification shifting. *GROWTH* and *BENCHMARK* also highlight a positive relationship, meaning that generally more profitable, growing, and leveraged companies are keener to engage in an activity of classification shifting. We can notice that there are significant relationships with all control variables tested in our analysis.

# [Table 4 near here]

Table 5a shows a negative relationship between  $UE\_CE$  and MASC with a rejection of our hypothesis 4. These results do not confirm the findings of Kim et al. (2017) in which the level of masculinity of the national culture influence positively the earnings management behaviours. In this case we must consider the peculiarity of our earnings management figure where the meaning of the manipulations is not to misreport the final income figure for a personal benefit but to manage core earnings often to meet investors' expectations. This result is in line with Hypothesis 4. Competitiveness, assertiveness, and individual achievement are characteristics of nations or communities that are thought to be dominated of male culture, with power and success seen as tangible rewards (Ho et al., 2012; Hofstede, 1980). The objectives in this cultural dimension are excellence and success. Therefore, higher scores for masculinity in a nation are typically associated with a management style that is motivated by ego, values money, and property, and favours strong economic growth over high pay. The research by Kim et al. (2017) provides evidence for the stronger impact of masculinity on managers' behaviours related to earnings manipulation. In our identification we also include control variables. To be concise and facilitate the presentation of results, we report the main variables. Results of control variables are available under request.

The significant relationship at the 1% level is still present for the whole sample and for the two subsamples between *LEV* and *UE\_CE*, confirming the results of DeFond and Jiambalvo (1994) about the possible manipulations put in place by managers to move earnings upwards to meet debt covenants or contracts. *MBV* is positive and significantly related to *UE\_CE* for the whole sample and WEU, suggesting that firms are generally less likely to engage in classification shifting when the book value is low.

Table 5b provides variability in the results of the relationship between *UE\_CE* and *IND*. For the EU28 and for WEU, there is a positive and significant impact from *IND* to *UE\_CE*, whereas for EEU the impact is negative confirming our second hypothesis. To this end, our findings for WEU confirm hypothesis 2 of positive association between individualism and classification shifting. However, for EEU there is a negative impact of individualism on classification shifting that confirms Han et al. (2010) arguing that the EEU are far less individualistic and self-oriented in terms of culture compared to the WEU. This can be explained by the different meaning of individualism in the EEU vis a vis the WEU, in which the "self" is still closely associated and identified with the "community". The former collectivist and the democratic centralist approaches of EEU have built up a concept of "individual interests" which are assimilated with the "group interests" and "mutual dependence".

Table 5c shows a negative relationship between *UE\_CE* and *PD* across all samples, confirm our first hypothesis. Our findings confirm the previous literature on the implication of a higher level of power distance in a country for earnings manipulation behaviour (Kim et al., 2017; Waldman et al., 2006). This result is in line with Hypothesis 1. Bakacsi et al. (2002), Kets de Vries (2000) and Schwartz & Bardi (2001) emphasise the predominance of hierarchical values in EEU, that explain that higher power distance would lead to lower classification shifting (Kim et al., 2017; Waldman et al., 2006). In general, managers have more legal and social pressures to be less opportunistic in countries with a higher power distance index, because they place less importance on their autonomy. For the WEU the managers' autonomy on the manipulations of companies' earnings has been inevitably reduced by the IFRS's application and business law and regulation. The same factors have an impact on the EEU countries, even more so because in the EEU countries there was no strong sense of "autonomy" due to the democratic centralism, which have enhanced interdependent behaviours.

Overall, results show that *PD* mitigates managers' incentive to misclassify revenue items upwards to increase reported core earnings.

Table 5d shows a positive relationship between  $UE\_CE$  and long-term orientation (*LTO*), which has not been thoroughly investigated. This result is in line with our expectation as discussed by Hypothesis 5. Considering the high level of variability of the capital markets in the aftermath of financial crises, managers could engage in classification shifting not just for serving their own opportunistic interests, but for assuring the long-term sustainability of their firms. One, of course, could argue that earnings management practises are justified in periods of financial turmoil, but it provides the intuition behind the positive association between  $UE\_CE$  and *LTO*. Clearly, in the long-term classification the shifting might prove that does little to the firm's sustainability. To study this type of effects we should collect longer time

series of both *UE\_CE* and *LTO*. Due to data limitations, we have no access to these at present. However, our results show the underlying complexities of the association between cultural values and earnings management.

Moreover, the long-term orientation represents a country that is more future-focused and values economic strength and perseverance (Halkos & Skouloudis, 2017; Hofstede et al., 2010). Contrarily, a short-term oriented society values tradition and upholds its social responsibilities with an emphasis on immediate results, which may include contentment and enjoyment in the here and now (Hofstede, 1994; Thanetsunthorn & Wuthisatian, 2018).

[Table 5a to 5d near here]

#### 5.4 Testing the impact of interaction between SPITEM and Culture Variables

Given the importance of SPITEM for classification shifting, next we investigate the impact of interactions between cultural variables and SPITEM on classification shifting. It could be the case that certain cultural variables would moderate or amplify the impact of SPITEM on classification shifting that has been reported to be positive.

Table 6 includes the interaction between *SPITEM* and *UAI* (*SPITEM* × *UAI*). The evidence varies. The impact of interaction *SPITEM* × *UAI* is positive, and significant at 5% level for the whole EU sample and for EEU, suggesting that *UAI* fails to moderate the effect of *SPITEM* on classification shifting. This is of interest because the individual impact of *UAI* on classification shifting is negative. However, the joint effect of *UAI* and *SPITEM* × *UAI* is negative. For the WEU the interaction SPITEM × UAI is insignificant.

[Table 6 near here]

Table 7 reports the interaction between *SPITEM* and *MASC* (*SPITEM*  $\times$  *MASC*). The evidence again varies in WEU vis a vis EEU. The negative effect of *MASC* on *UE\_CE* is amplified in the presence of *SPITEM* in EEU though the magnitude is small, while for the WEU the results are not significant.

# [Table 7 near here]

Table 8 contains the interaction between *SPITEM* and *IND* (*SPITEM*  $\times$  *IND*). The interaction term is insignificant for the WEU sub sample, whereas for the EEU *IND* fails to moderate the impact of SPITEM as seen in the case of *UAI*, though the joint effect of *IND* and *SPITEM*  $\times$  *IND* is negative.

### [Table 8 near here]

Table 9 contains the interaction between *SPITEM* and *PD* (*SPITEM*  $\times$  *PD*). Results are of interest as the effect of *SPITEM*  $\times$  *PD* is negative and significant at 5% level for the EU28, but positive for the sub-samples. These results highlight the complexities of the underlying relationships and provide evidence of the underlying heterogeneity across the different samples. The underlying heterogeneity of EU across its Member States could highlight the fact that there is no 'one-size-fits-all' main finding. In the case of cultural differences (with heterogeneity across countries) *PD* is present and feeds into our empirical findings. There should be a dominant Member State at the EU level that drives the negative effect of *SPITEM*  $\times$  *PD* on classification shifting, which is not feasible to explore in the current context due to data limitations. Note, however, that the joint effect of *PD* and *SPITEM*  $\times$  *PD* is positive across all samples.

#### [Table 9 near here]

Table 10 reports the interaction between *SPITEM* and *LTO* (*SPITEM* × *LTO*). Again, we find variability between WEU and EEU. For the WEU, findings show a positive and significant relationship at 5% level between the interaction of *SPITEM* × *LTO* and *UE\_CE*, while for EEU this relationship is reported negative and significant (at 5% level), suggesting that *LTO* would mitigate the impact of *SPITEM* on *UE\_CE*. Yet, the joint effect of *LTO* and *SPITEM* ×*LTO* on classification shifting is positive.

## [Table 10 near here]

#### 6. Real earnings management and Hofstede - Cultural Variables

As a robustness check, this study also considers the relationship between *REM1*, *REM2* and all the five WVS questions that capture people's values. It is worth noting that the second measure of REM *REM2* is obtained using abnormal cash flows and abnormal discretionary expenses to consider criticism of measurement errors in *REM1* (see Cohen & Zarowin 2010; Roychowdhury, 2006; Zang, 2012).

Results regarding *REM1* and *REM2* are reported in Tables 11 and 12 respectively. All in all, parameter estimates of cultural variables are in line with the ones reported for the case of classification shifting. Similarly, control variables show similar effects as seen in the classification shifting models. In some detail, it is worth noting (see Table 11) that IND has a larger magnitude impact *REM1* in the case of EEU vis a vis WEU, whereas PD seems to subdue *REM1* in EEU while it asserts a positive impact on WEU. These results might not come as a surprise, as EEU of EU have been under a centrally planned economy structure for decades,

and cultural variables like PD would be expected to have a significant impact from an economic and statistical point of view. Similarly, patterns are observed for the long-term orientation, see LTO, where its impact is much more pronounced for the EEU compared to the WEU. However, MASC appears to assert a negative and significant impact on REM1 in both sub-samples, WEU and EEU.

# [Table 11 near here]

Table 12 reports results for the second measure of REM, *REM2*. Results show variability, suggesting the underlying complexities of cultural variables across Member States. In some detail, in the case of *REM2*, MASC asserts a negative impact on EEU compared to a positive impact on WEU, whilst PD asserts a positive impact on EEU whereas it has a negative impact on WEU.

#### [Table 12 near here]

#### 7. Conclusion

Checking whether the EEU countries are engaging in classification shifting, our results show that these countries are manipulating the core earnings with a higher magnitude compared to the WEU. To this end, EEU is catching up to the financial misreporting of their colleagues in WEU, although the changes in institutional factors, such as harmonization and application of accounting regulation and investment protection, the shift from a closed to an open capital market as well as an approach to the maximization of the profit happened only two decades ago. This first result seems to support the view the European Union integration process has

been relatively effective, though according to Akaliyski (2019) such process has also altered values and beliefs in the EEU.

On this assumption, we have also examined the effects of Hofstede 'cultural dimensions (individualism, uncertainty avoidance, power distance, masculinity, and long-term orientation). Our findings show that cultural factors play a significant role at the European level and confirm that there are still differences in the two sub-samples considered. In addition, our results show there is no 'one-size-fits-all' case as there is variability. The cultural dimensions aligned between the two blocs of countries are Power distance, Masculinity and Long-Term orientation, however only the first confirm its negative relationships with earnings management in line with our hypothesis and with the literature (Kim et al., 2017). Individualism and Uncertainty avoidance, present different effects on the classification shifting behaviour applied in the WEU and EEU, confirming that in the EEU the sense of "community" and the collective culture ("obshcina") still play a significant role in mitigating classification shifting although the IFRS and other regulations have been generally applied. The findings are important for regulators, external monitors, and investors. They show how cultural dimensions of the EEU should be taken into account in planning future action for enhancing financial reporting quality across the EU. Future research could explore how EU accounting and finance regulations have been embraced across EU Member States and further examined whether EEU cultural dimensions have played a role. Additionally, future investigations might take into account the cultural implication of the recognition of the EU legitimacy in EEU or/and how the presence of Western companies operating in EEU have been facilitating the swift in the cultural dimensions and classification shifting practices.

This study is limited in the analysis of the timing in which the EEU have joined the European Union. The effects of cultural dimensions on classification shifting can be influenced by how long each Eastern country has been influenced by the integration process. Additionally, <text>

#### 

#### References

- Acemoglu, D., Johnson, S., & Robinson, J. (2005). The Rise of Europe: Atlantic Trade, Institutional Change, and Economic Growth. *American Economic Review*, 95(3), 546– 579. <u>https://doi:10.1257/0002828054201305</u>
- Ahern, K., Daminelli, D., & Fracassi, C. (2015). Lost in translation? The effect of cultural values on mergers around the world. *Journal of Financial Economics*, 117, 165–189. <u>https://10.1016/j.jfineco.2012.08.006</u>
- Akaliyski, P. (2019). United in diversity? The convergence of cultural values among EU member states and candidates. *European Journal of Political Research*, 58, 388–411. <u>https://doi.org/10.1111/1475-6765.12285</u>
- Albu, C., Albu, C. T., & Gray, S. J. (2020). Institutional factors and the impact of international financial reporting standards: the Central and Eastern European experience. *Accounting Forum*, 44(3), 184–214. <u>https://doi.org/10.1080/01559982.2019.1701793</u>
- Ashbaugh, H., Lafond, R., & Mayhew, B. (2003). Do non-audit services compromise auditor independence? Further Evidence. *The Accounting Review*. 78(3), 611–640. https://doi.org/10.2308/accr.2003.78.3.611
- Athanasakou, V. E., Strong, N. C., & Walker, M. (2009). Earnings Management or Forecast Guidance to Meet Analyst Expectations? *Accounting and Business Research*, 39(1), 3–35. <u>https://doi.org/10.1080/00014788.2009.9663347</u>
- Bakacsi, G., Takacs, S., Karacsonyi, A., & Imrek, V. (2002). East European Cluster: Tradition and Transition. *Journal of World Business*, 37(1), 69–80. <u>https://doi.org/10.1016/S1090-9516(01)00075-X</u>
- Barton, J., & Simko, P. J. (2002). The balance sheet as an earnings management constraint. *The Accounting Review*, 77(s-1), 1–27. <u>https://doi.org/10.2308/accr.2002.77.s-1.1</u>

- Behn, B. K., Gotti, G., Herrmann, D., & Kang, T. (2013). Classification Shifting in an International Setting: Investor Protection and Financial Analysts Monitoring. *Journal of International Accounting Research*, 12(2), 27–50. https://doi.org/10.2308/jiar-50439
- Bermpei, T., Kalyvas, A. N., Neri., L., & Russo, A. (2022). Does Economic Policy Uncertainty Matter for Financial Reporting Quality? Evidence from the United States. *Review of Quantitative Finance and Accounting*, 58(2), 795–845. <u>https://doi.org/10.1007/s11156-</u> 021-01010-2
- Beugelsdijk, S., & Welzel, C. (2018). Dimensions and dynamics of national culture: synthesizing Hofstede with Inglehart. *Journal of Cross-Cultural Psychology*, 49, 1469– 1505. <u>https://doi.org/10.1177/0022022118798505</u>
- Bhattacharya, N., Black, E. L., Christensen, T. E., & Mergenthaler, R. D. (2007). Who trades on pro-forma Earnings Information? *The Accounting Review*, 82(3), 581–619. <u>https://doi.org/10.2308/accr.2007.82.3.581</u>
- Boahen, E. O. (2017). The Impact of Religiosity, Culture, Legal Environment and Corporate Governance on Earnings Management Methods, University of Sussex.
- Boahen, E. O., & Mamatzakis, E. C. (2020). The impact of religion on classification shifting in the presence of corporate governance and BIG 4 audit. *Accounting Forum*, 44(2), 103–131. https://doi.org/10.1080/01559982.2019.1573404
- Boahen, E. O., & Mamatzakis, E. C. (2021). What are the effects of culture and institutions on classification shifting in India? *Journal of International Accounting, Auditing and Taxation*, 44, 1–22. <u>https://doi.org/10.1016/j.intaccaudtax.2021.100402</u>
- Brochet, F., Miller, G., Naranjo, P. L., & Yu, G. (2019). Managers' Cultural Background and Disclosure Attributes. *The Accounting Review*, 94(3), 57–86. https://doi.org/10.2308/accr-52290

- Callen, J. L., Morel, M., & Richardson, G. (2011). Do culture and religion mitigate earnings management? Evidence from a cross-country analysis. *International Journal of Disclosure and Governance*, 8(2), 103–121. <u>https://doi: 10.1057/jdg.2010.31</u>
- Cheng, Q., & Warfield, T. D. (2005). Equity incentives and earnings management. *The Accounting Review*, 80(2), 441–476. <u>https://doi.org/10.2308/accr.2005.80.2.441</u>
- Cohen, D., Pandit, S., Wasley, C. E., & Zach, T. (2020). Measuring real activity management. *Contemporary Accounting Research*, 37(2), 1172–1198. <u>https://doi.org/10.1111/1911-</u> <u>3846.12553</u>
- Cohen, D. A., & Zarowin, P. (2010). Accrual-based and real earnings management activities around seasoned equity offerings. *Journal of Accounting and Economics*, 50(1), 2–19. <u>https://doi.org/10.1016/j.jacceco.2010.01.002</u>
- Davis, L. S., & Abdurazokzoda, F. (2016). Language, Culture, and institutions: Evidence from a new linguistic dataset. *Journal of Comparative Economics*, 44(3), 541–561. <u>https://doi.org/10.1016/j.jce.2015.10.015</u>
- Dechow, P., Sloan, R., & Sweeney, A. (1996). Causes and consequences of earnings manipulation: An analysis of firms' subject to enforcement actions by the SEC. *Contemporary Accounting Research*, 13(1), 1–36. <u>https://doi.org/10.1111/j.1911-3846.1996.tb00489.x</u>
- Defond, M., & Jiambalvo, J. (1994). Debt covenant violation and manipulation of accruals. Journal of Accounting and Economics. 17, 145–176. <u>https://doi.org/10.1016/0165-4101(94)90008-6</u>
- Desender, K. A., Castro, C. E., & De Leon, S. A. E. (2011). Earnings Management and Cultural Values. *The American Journal of Economics and Sociology*, 70(3), 639–670. https://doi.org/10.1111/j.1536-7150.2011.00786.x

- Doupnik, T. S., & Tsakumis, G. T. (2004). A Critical Review of the Tests of Gray's Theory of Cultural Relevance and Suggestions for Future Research. *Journal of Accounting Literature*, 23(1), 1–30. https://doi.org/216304608
- Doupnik, T. S. (2008). Influence of Culture on Earnings Management: A Note. *Abacus*, 44(3), 317–340. <u>http://doi:10.1111/j.1467-6281.2008.00265.x</u>.
- Doyle, J., Lundholm, R., & Soliman, M. (2003). The predictive value of expenses excluded from pro forma earnings. *Review of Accounting Studies*, 8(2-3), 145–174. <u>http://dx.doi.org/10.1023/A:1024472210359</u>
- Eder, A., Volk, A., & Haller, M. (2017). Patterns of Social Ties, Trust, and Participation after the Fall of the Iron Curtain: New Findings from Central and Southeast European Countries. *International Journal of Sociology*, 47(3), 147–161. <u>https://doi.org/10.1080/00207659.2017.1335524</u>
- Fama, E. F., & French, K. R. (1997). Industry of cost of Equity. Journal of Financial Economics, 44, 153–193. <u>https://doi.org/10.1016/S0304-405X(96)00896-3</u>
- Fan, Y., Barua, A., Cready, W. M., & Thomas, W. (2010). Managing Earnings Using Classification Shifting: Evidence from Quarterly Special Items. *The Accounting Review*, 85(4), 1303–1323. <u>https://doi.org/10.2308/accr.2010.85.4.1303</u>
- Fan, J. O., Rui, M., & Zhao, M. (2008). Public governance and corporate finance: Evidence from corruption cases. *Journal of Comparative Economics*, 36, 343–364. <u>https://doi.org/10.1016/j.jce.2008.05.001</u>
- Fernandes, N., & Ferreira, M. A. (2007). The Evolution of Earnings Management and Firm Valuation: A Cross-country Analysis. Universidade Catolica Portugeusa. Working Paper, October.
- Fidrmuc, J. P., & Jacob, M. (2010). Culture, Agency Cost and Dividends. *Journal of Comparative Economics*, 38(3), 321–339. <u>https://doi.org/10.1016/j.jce.2010.04.002</u>

- Francis, J., & Wang, D. (2008). The joint effect of investor protection and Big 4 audits on earnings quality around the world. *Contemporary Accounting Research*, 25(1), 157– 191. https://doi.org/10.1506/car.25.1.6
- Gray, S. (1988). Towards a theory of cultural influence on the development of accounting systems internationally. *Abacus*, 24(1), 1–15. <u>https://doi.org/10.1111/j.1467-6281.1988.tb00200.x</u>
- Guan, L., Pourjalali, H., Sengupta, P., & Teruya, J. (2005). Effect of cultural environment on earnings manipulation: A five Asia-Pacific country analysis. *Multinational Business Review*, 13(2), 23–41. <u>https://doi.org/10.1108/1525383X200500007</u>
- Guo, H., & Jiang, X. (2011). Accruals and the Conditional Equity Premium. Journal of Accounting Research, 49, 187–221. <u>https://doi.10.1111.1475-679x.2010.00393.x</u>
- Haga, J., Hoglunda, H., & Sundvika, D. (2018). Stock market listing status and real earnings management. *Journal of Accounting and Public Policy*, 37(5), 420–435. <u>https://doi.org/10.1016/j.jaccpubpol.2018.09.002</u>
- Halkos, G. & Skouloudis, A. (2017). Revisiting the relationship between corporate social responsibility and national culture: a quantitative assessment. *Management Decision*, 55(3), 595–613. <u>https://doi.org/10.1108/MD-12-2016-0868</u>
- Han, S., Kang, T., Salter, S., & Yoo, Y. K. (2010). A cross-country study on the effects of national culture on earnings discretion. *Journal of International Business Studies*, 41, 123–142. <u>https://doi.org/10.1108/IJOEM-09-2019-0734</u>
- Haselmann, R., Pistor, K., & Vig, V. (2010). How Law Affects Lending. *Review of Financial Studies*, 23(2), 549–580. <u>https://doi.org/10.1093/rfs/hhp073</u>
- Haw, M., Ho, S. M., & Li, A. Y. (2011). Corporate Governance and Earnings Management by Classification Shifting. *Contemporary Accounting Research*, 28, 517–553.
   <u>https://doi.org/10.1111/j.1911-3846.2010.01059.x</u>

- Healy, P. & Wahlen, J. (1999). A review of the earnings management literature and its implications for standard setting. *Accounting Horizons*, 13(4), 365–383. https://doi.org/10.2308/acch.1999.13.4.365
- Hirshleifer, D., Hou, K., & Teoh, S. (2009) Accruals, Cash Flows, and Aggregate Stock Returns. *Journal of Financial Economics*, 91, 389–406. https://doi.org/10.1016/j.jfineco.2007.11.009
- Ho, F. N., Wang H. M. D., & Vitell, S. J. (2012). A global analysis of corporate social performance: the effect of cultural and geographic environment. *Journal of Business Ethics*, 107(4), 423–433. <u>https://doi.org/10.1007/S10551-011-1047-Y</u>
- Hofstede, G. (1980). *Culture's Consequences: International Differences in Work-Related Values*, 2<sup>nd</sup> edition. Sage Publications.
- Hofstede, G. (1991) *Cultures and Organizations: Software of the Mind*. New York: McGraw-Hill.
- Hofstede, G. (1994). The business of international business is culture. *International Business Review*, 3(1), 1–14. <u>https://doi.org/10.1016/0969-5931(94)90011-6</u>
- Hofstede, G. (2001). *Culture's Consequences: Comparing Values, Behaviours, Institutions and Organizations across Nations*, 2<sup>nd</sup> edition. Sage Publications.
- Hofstede, G., Hofstede, G., & Minkov, M. (2010). *Cultures and Organizations: Software of the Mind*, 3<sup>rd</sup> edition. McGraw-Hill.
- Hutchings, K., & Michailova, S. (2006). The impact of group membership on knowledge sharing in Russia and China. *International Journal of Emerging Markets*, 1(1), 21–34. <u>https://doi/10.1108/17468800610644988</u>
- Jackson, A. B. (2018). Discretionary accruals, earnings management ... or not? *Abacus*, 54(2), 136–153. https://doi.org/10.1111/abac.12117

- Jarvinen, T., & Myllymaki, E. R. (2016). Real Earnings Management before and after reporting SOX 404 Material weaknesses. *Accounting Horizon*, 30(1), 119–141. https://doi.org/10.2308/acch-51310
- Kets de Vries, M. F. R. (2000). A journey into the "Wild East:" Leadership style and organizational practices in Russia. Organizational Dynamics, 28(4), 67–81. <u>https://doi.org/10.1016/j.orgdyn.2008.04.002</u>
- Kim, J., Kim, Y., & Zhou, J. (2017). Languages and earnings management. Journal of Accounting and Economics, 63(2-3), 288–306. <u>https://doi.org/10.1016/j.jacceco.2017.04.001</u>
- Kim, Y., & Park, M. S. (2014). Real Activities Manipulation and Auditors' Client-Retention Decisions. *The Accounting Review*, 89(1), 367–401. <u>https://doi.org/10.2308/accr-50586</u>
- Kirkman, B., Lowe, K., & Gibson, C. (2006). A quarter century of Culture's consequences: A review of empirical research incorporating Hofstede's cultural values framework. *Journal of International Business Studies*, 37(3), 285–320. <u>https://doi.org/10.1057/palgrave.jibs.8400202</u>
- Klasing, M. J. (2013). Cultural dimensions, collective values, and their importance for institutions. *Journal of Comparative Economics*. 41, 447–467. <u>https://doi.org/10.1016/j.jce.2012.09.003</u>
- Knechel, W. R., Thomas, E., & Driskill, M. (2020). Understanding financial auditing from a service perspective. Accounting, Organizations and Society, 81, 2–23. <u>https://doi.org/10.1016/j.aos.2019.101080</u>
- Kolman, L., Noorderhaven, N., Hofstede, G., & Dienes, E. (2003). Cross-cultural differences in Central Europe. *Journal of Managerial Psychology*, 18(1), 76–88. <u>https://doi.org/10.1108/02683940310459600</u>

- Kothari, S. P., Leone, A. J., & Wasley, C. E. (2005). Performance Matched Discretionary Accruals Measures. *Journal of Accounting and Economics*, 39, 163–197. https://doi.org/10.1016/j.jacceco.2004.11.002
- Kothari, S. P., Mizik, N., & Roychowdhury, S (2016). Managing for the Moment. The Role of Earnings Management via Real Activities vrs Accruals in SEO Valuation. *The Accounting Review*, 91(2), 559–586. <u>https://doi.org/10.2308/accr-51153</u>

Kuchma, L. (2003). YKpaulla - 11e PoccuR. [The Ukraine is not Russia]. Moscow: Vremia.

- Kury, K. W. (2007). Decoupled earnings: An institutional perspective of the consequences of maximizing shareholder value. *Accounting Forum*, 31(4), 370–383. <u>https://doi.org/10.1016/j.accfor.2007.09.003</u>
- Leuz, C., Nanda, D., & Wysocki, P. (2003). Earnings Management and Investor Protection: An International Comparison. *Journal of Financial Economics*, 69, 125–149. <u>https://doi.org/10.1016/S0304-405X(03)00121-1</u>
- Li, K., Griffin, D., Yue, H., & Zhao, L. (2013). How does culture influence Corporate Governance Risk-Taking? *Journal of Corporate Finance*, 23, 1–22. https://doi.org/10.1016/j.jcorpfin.2013.07.008
- Maslow, A. H. (1943). A theory of human motivation. *Psychological Review*, 50(4), 370–396. https://doi.org/10.1037/h0054346
- McVay, S. E. (2006). Earnings Management using Classification shifting: an examination of core earnings and special items. *The Accounting Review*, 81(3), 501–531. <u>https://doi.org/10.2308/accr.2006.81.3.501</u>
- Mueller, D. C., & Peev, E. (2007). Corporate governance and investment in Central and Eastern Europe. *Journal of Comparative Economics*, 35, 414–437. https://doi.org/10.1016/j.jce.2007.02.001

- Myers, J. N., Myers, L. A., & Skinner, D. J. (2007). Earnings Momentum and Earnings Management. *Journal of Accounting, Auditing & Finance*, 22(2), 249–284. https://doi.org/10.1177/0148558X0702200211
- Nguyen, L.A., Dellaportas, S., Vesty, G.M., Pham, V.A.T., Jandug, L., & Tsahuridu, E. (2022). The influence of organisational culture on corporate accountants' ethical judgement and ethical intention in Vietnam, *Accounting, Auditing & Accountability Journal*, 35(2), 325–354.
- Nelson, M. W., Elliott, J., Tarpley, R. L., & Gibbins, M. (2002). Evidence from Auditors about Managers' and Auditors' Earnings Management Decisions. *The Accounting Review*, 77, 175–202. <u>https://doi.org/10.2308/accr.2002.77.s-1.175</u>
- Nissim, D., & Penman, S. H. (2001). Ratio Analysis and Equity Valuation: From Research to Practice. *Review of Accounting Studies*, 6(1), 109–154. <u>https://doi.org/10.1023/A:1011338221623</u>
- Paredes, A. A. P., & Wheatley, C. (2017). The influence of culture on real earnings management. *International Journal of Emerging Markets*, 12(1), 38–57. https://doi.org/10.1108/IJoEM-12-2014-0218
- Persson, T., & Tabellini, G. (2009). Democratic Capital: The Nexus of Political and Economic Change. *American Economic Journal: Macroeconomics*, 1(2), 88-126. DOI: 10.1257/mac.1.2.88
- Ringov, D., & Zollo, M. (2007). Corporate responsibility from a socio-institutional perspective: the impact of national culture on corporate social performance. *Corporate Governance*, 7(4), 476–485. https://doi.org/10.1108/14720700710820551

Roychowdhury, S. (2006). Earning Management through real activities manipulation. Journal

of Accounting and Economics, 42(3), 335–370. https://doi.org/10.1016/j.jacceco.2006.01.002

- Schmidt, V. A. (2019). Politicization in the EU: between national politics and EU political dynamics. *Journal of European Public Policy*, 26(7), 1018–1036. https://doi.org/10.1080/13501763.2019.1619189
- Schwartz, S. H., & Bardi, A. (2001). Value hierarchies across culture: Taking a similarities perspective. *Journal of Cross-Cultural Psychology*, 32(3), 268–290. https://doi.org/10.1177/0022022101032003002
- Stiglitz, J. E. (1989). On the Economic Role of the State, in: A. Heertje (Ed.), *The Economic Role of the State* (pp. 9–85). Basil Blackwell.
- Stulz, R., & Williamson, R. (2003). Culture, Openness and Finance. Journal of Financial Economics, 70(3), 313–349. <u>https://doi.org/10.1016/S0304-405X(03)00173-9</u>
- Sun, J., & Lan, G. (2014). Independent Audit Committee Characteristics and Real Earnings Management. Managerial Auditing Journal, 29(2), 153–172. . <u>https://doi.org/10.1108/MAJ-05-2013-0865</u>
- Tang, L., & Koveos, P. E. (2008). A framework to update Hofstede's cultural value indices: Economic dynamics and institutional stability. *Journal of International Business Studies*, 39, 1045–1063. <u>https://doi.org/10.1057/palgrave.jibs.8400399</u>
- Thanetsunthorn, N., & Wuthisatian, R. (2018). Cultural Configuration Models: Corporate Social Responsibility and National Culture. *Management Research Review*, 41(10), 1137–1175. <u>https://doi.org/10.1108/MRR-08-2017-0254</u>
- Varnum, M. E. W., Grossmann, I., Katunar, D., Nisbett, R. E., & Kitayama, S. (2008). Holism in a European cultural context: Differences in cognitive style between Central and East Europeans and Westerners. *Journal of Cognition and Culture*, 8(3-4), 321–333. https://doi.org/10.1163/156853708X358209
- Waldman, D. A., De Luque, M. S., Washburn, N., House, R. J., Adetoun, B., Barrasa, A., & Wilderom, C. P. (2006). Cultural and leadership predictors of corporate social

1	
2	
4	
5	
6	
7	
8	
9	
10	
12	
13	
14	
15	
16	
17	
18	
20	
21	
22	
23	
24	
25	
20 27	
28	
29	
30	
31	
32	
33 34	
35	
36	
37	
38	
39	
40 41	
42	
43	
44	
45	
46	
47 48	
49	
50	
51	
52	
53	
54 55	
56	
57	

responsibility values of top management: A GLOBE study of 15 countries. Journal of International **Business** Studies, 37, 823-837. https://doi.org/10.1057/palgrave.jibs.8400230

- Zalata, A., & Robert, C. (2015). Internal Corporate Governance and Classification Shifting Practices - An Analysis of U.K. Corporate Behaviour. Journal of Accounting, Auditing and Finance, 31(1), 51-78. https://doi.org/10.1177/0148558X15571736
- Zang, A. Y. (2012). Evidence on the Trade off between Real Manipulation and Accrual Manipulation. The Accounting Review, 87(2), 675–703. https://doi.org/10.2308/accr-

27. 70 F

Abbreviations	Meaning
ITAI	Generally speaking would you say that most people can be
0/11	trusted or that you need to be very careful in dealing with
7	neonle?
C	People.
MASC	Men should have more right to a job than women
IND	Incomes should be made more equal
PD	Private ownership of business should be increased
LTO	National pride
	46
	Abbreviations UAI MASC IND D LTO

# Table 2. Descriptive statistics.

5			WEU			
Variable	Obs	Mean	Std. dev.	Min	Max	
UE_CE	68,200	-0.950636	1.334846	-1.55563	0.8682073	
REM1	68,200	0.2513372	0.6171511	-1.914479	3.291414	
REM2	68,200	-0.3850883	0.232891	-1.570834	-0.2061559	
SPITEM	68,200	0.0166673	0.045743	0	0.2257756	
UAI	68,200	0.6681526	0.1433022	0.32	0.92	
MASC	68,200	0.7956343	0.0927514	0.34	0.98	
ND	68,200	5.081061	0.7969354	0	6.09	
PD	68,200	4.90331	0.2585417	4.16	5.54	
LTO	68,200	1.753236	0.2632681	1.31	2.4	
ROA	68,200	0.0065401	0.2071599	-1.050746	0.4030962	
SIZE	68,200	5.383291	2.903174	-0.8698844	13.85758	
MBV	68,200	3.837095	6.693637	-10.37798	44.01131	
LEV	68,200	0.5484592	0.3164304	0.0048465	1.877104	
GROWTH	68,200	0.1026817	0.3222051	-0.4872325	0.990605	
EXPOSURE	68,200	0.255761	0.2854782	-0.0950299	1.359554	
		EE	U			
Variable	Obs	Mean	Std. dev.	Min	Max	
UE CE	11,080	-0.847092	1.306465	-1.33273	1.578676	
REM1	11,080	0.2561604	0.6260299	-1.783286	3.288493	
REM2	11,080	-0.4009217	0.265223	-1.570834	-0.2061407	
SPITEM	11,080	0.0149667	0.0446274	0	0.2257756	
UAI	11,080	0.7862561	0.0438927	0.71	0.93	
MASC	11,080	0.6300856	0.0937571	0.43	0.89	
IND	11,080	6.028114	0.7742489	3.48	7.61	
PD	11,080	5.837713	0.9360385	3.29	6.75	
UAI	11,080	1.584278	0.2205169	1.34	2.22	
ROA	11,080	0.0290461	0.1602962	-1.050746	0.4030962	
SIZE	11,080	5.15729	2.851859	-0.8698844	13.85758	
MBV	11,080	2.867603	4.473934	-10.37798	44.01131	
LEV	11,080	0.5109883	0.3226684	0.0048465	1.877104	
GROWTH	11,080	0.0996803	0.3344038	-0.4872325	0.990605	
EXPOSURE	11,080	0.1759752	0.2411788	-0.0950299	1.359554	
			47			

	(1) EU28	(2) WFU	(3) FEU	
VARIABLES	UE CE	UE CE	UE CE	
SPITEM	0 205***	0.287***	0 476***	
	(4,193)	(3.497)	(3.815)	
Constant	0.060***	0.053***	0.106***	
	(31.363)	(24.988)	(22.419)	
	~ - )		× /	
Observations	79,570	68,200	11,080	
R-squared	0.004	0.006	0.001	
Number of Firms	8,401	7,160	1,221	
Industry FE	YES	YES	YES	
Year FE	YES	YES	YES	
Country FE	YES	YES	YES	
Note: t-statistics in parentl	neses, *** p<0.01, *	** p<0.05, * p<0.	1	
rote. e statistics in parenti	10505, p 0.01,	p 10.00, p 10.	1	
		48		

# Table 3. Evidence of Classification shifting and special items.

Page 49 of 60

1

2	
2	
5	
4	
F	
С	
6	
-	
/	
8	
0	
9	
10	
10	
11	
12	
12	
13	
11	
14	
15	
10	
10	
17	
10	
18	
19	
~~	
20	
21	
~ 1	
22	
25	
25	
24	
25	
25	
26	
27	
27	
28	
20	
29	
30	
21	
31	
32	
52	
33	
34	
5-	
35	
36	
50	
37	
20	
20	
39	
10	
40	
41	
<b>⊿</b> ∩	
42	
43	
11	
44	
45	
10	
46	
47	
40	
48	
49	
50	
51	
51	
52	
52	
22	
54	
55	
22	
56	
J/	

58 59 60

# Table 4. Classification shifting and uncertainty avoidance (UAI).

	(1) EU28	(2) WEU	(3) EEU	
VARIABLES	UE_CE	UE_CE	UE_CE	
UAI	0.242***	0.245***	-0.061**	
	(3.208)	(3.148)	(-2.084)	
SPITEM	0.272***	0.204***	0.552***	
	(9.329)	(8.028)	(4.656)	
ROA	0.650***	0.568***	1.161***	
	(9.762)	(7.879)	(6.477)	
SIZE	0.069***	0.067***	0.090***	
	(12.218)	(11.021)	(5.383)	
MBV	0.006***	0.006***	0.005	
	(4 871)	(4 673)	(1.482)	
LEV	1 469***	1 415***	1 707***	
	(41 434)	(35,819)	(20,767)	
GROWTH	1 401***	1 443***	1 189***	
Gite will	(73, 383)	(68 454)	(25.847)	
EXPOSURE	-0 625***	-0 645***	-0 451***	
EAIOSOKE	(21,079)	(10.048)	(5,875)	
Constant	2 582***	2 561***	(-5.675)	
Collstallt	(22.082)	(22.659)	(5,960)	
	(-23.982)	(-22.038)	(-3.809)	
Observations	A1 A61	21002	6 410	
Observations	41,401	54,885	0,419	
R-squared	0.169	0.169	0.172	
Number of Firms	7,014	5,906	1,089	
Industry FE	YES	YES	YES	
Year FE	YES	YES	YES	
Country FE	YES	YES	YES	
Note: t-statistics in parenthe	ses, *** p<0.01, ** p<0.0	05, * p<0.1		
	49			

~	(1) EU28	(2) WEU	(3) EEU
VARIABLES	UE_CE	UE_CE	UE_CE
MASC	-0.628***	-0.858***	-0.124***
	(-4.144)	(-4.644)	(-3.452)
SPITEM	0.290***	0.236***	0.554***
	(9.460)	(8.235)	(4.666)
Constant	-2.962***	-2.760***	-3.456***
	(-25.555)	(-19.291)	(-17.729)
Observations	41,461	34,883	6,419
R-squared	0.170	0.170	0.172
Number of Firms	7,014	5,906	1,089
Industry FE	YES	YES	YES
Year FE	YES	YES	YES
Country FE	YES	YES	YES

# Table 5a. Classification shifting and masculinity (MASC).

# Table 5b. Classification shifting and individualism (IND).

	(1) EU28	(2) WEU	(3) EEU
VARIABLES	UE_CE	UE_CE	UE_CE
IND	0.209**	0.107***	-0.117***
	(2.002)	(2.734)	(-2.858)
SPITEM	0.269***	0.201***	0.516***
	(9.309)	(8.014)	(4.553)
Constant	-3.409***	-3.420***	-2.845***
	(-57.095)	(-54.777)	(-10.752)
Observations	11 161	24.002	6 410
Observations	41,401	54,885	0,419
R-squared	0.169	0.169	0.174
Number of Firms	7,014	5,906	1,089
Industry FE	YES	YES	YES
Year FE	YES	YES	YES
Country FE	YES	YES	YES

# Table 5c. Classification shifting and power distance (PD).

	(1) EU28	(2) WEU	(3) EEU
VARIABLES	UE_CE	UE_CE	UE_CE
PD	-0.175***	-0.239***	-0.045***
	(-4.356)	(-4.612)	(-3.683)
SPITEM	0.573***	0.593***	0.542***
	(9.020)	(7.729)	(4.627)
Constant	-2.517***	-2.202***	-3.262***
	(-11.980)	(-8.575)	(-8.027)
Observations	28,625	22,047	6,419
R-squared	0.169	0.168	0.172
Number of Firms	5,288	4,180	1,089

Yx.

Industry FE	YES	YES	YES
Year FE	YES	YES	YES
Country FE	YES	YES	YES

### Table 5d. Classification shifting and long-term orientation (LTO).

	(1) EU28	(2) WEU	(3) EEU
VARIABLES	UE_CE	UE_CE	UE_CE
LTO	0.463***	0.430***	0.233*
	(5.242)	(4.776)	(1.901)
SPITEM	0.343***	0.303***	0.541***
	(9.146)	(7.919)	(4.627)
Constant	-4.395***	-4.350***	-5.018***
	(-27.238)	(-25.866)	(-6.338)
Observations	35.063	28 185	6 / 10
R-squared	0 172	0 172	0,173
Number of Firms	6.172	5 064	1 089
Industry FE	YES	YES	YES
Year FE	YES	YES	YES
Country FE	YES	YES	YES

Note: Note that we include the same control variables as in Table 4, namely ROA; SIZE; MBV; LEV; GROWTH; EXPOSURE. Results show high significance in parameter estimates and are available under request. t-statistics in parentheses, \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

in Tab, ificance in p<0.01, \*\* p<

VARIABLES SPITEM UAI SPITEM×UAI ROA SIZE MBV LEV GROWTH EXPOSURE	(1) EU28 UE_CE 0.244*** (3.363) 0.235** (2.168) 0.0321** (3.302) 0.650*** (9.755) 0.069*** (12.219) 0.006*** (4.878) 1.469*** (41.338) 1.401***	(2) WEU UE_CE 0.457*** (2.856) 0.253** (2.177) -0.364 (-0.329) 0.568*** (7.881) 0.067*** (11.021) 0.006*** (4.659) 1.416***	(3) EEU UE_CE 0.791*** (3.902) -0.191** (-2.160) 0.069** (2.137) 1.156*** (6.446) 0.090*** (5.388) 0.005
VARIABLES SPITEM UAI SPITEM×UAI ROA SIZE MBV LEV GROWTH EXPOSURE	UE_CE 0.244*** (3.363) 0.235** (2.168) 0.0321** (3.302) 0.650*** (9.755) 0.069*** (12.219) 0.006*** (4.878) 1.469*** (41.338) 1.401***	UE_CE 0.457*** (2.856) 0.253** (2.177) -0.364 (-0.329) 0.568*** (7.881) 0.067*** (11.021) 0.006*** (4.659) 1.416***	UE_CE 0.791*** (3.902) -0.191** (-2.160) 0.069** (2.137) 1.156*** (6.446) 0.090*** (5.388) 0.005
SPITEM UAI SPITEM×UAI ROA SIZE MBV LEV GROWTH EXPOSURE	0.244*** (3.363) 0.235** (2.168) 0.0321** (3.302) 0.650*** (9.755) 0.069*** (12.219) 0.006*** (4.878) 1.469*** (41.338)	$\begin{array}{c} 0.457^{***} \\ (2.856) \\ 0.253^{**} \\ (2.177) \\ -0.364 \\ (-0.329) \\ 0.568^{***} \\ (7.881) \\ 0.067^{***} \\ (11.021) \\ 0.006^{***} \\ (4.659) \\ 1.416^{***} \end{array}$	0.791*** (3.902) -0.191** (-2.160) 0.069** (2.137) 1.156*** (6.446) 0.090*** (5.388) 0.005
UAI SPITEM×UAI ROA SIZE MBV LEV GROWTH EXPOSURE	(3.363) 0.235** (2.168) 0.0321** (3.302) 0.650*** (9.755) 0.069*** (12.219) 0.006*** (4.878) 1.469*** (41.338)	$\begin{array}{c} (2.856) \\ 0.253^{**} \\ (2.177) \\ -0.364 \\ (-0.329) \\ 0.568^{***} \\ (7.881) \\ 0.067^{***} \\ (11.021) \\ 0.006^{***} \\ (4.659) \\ 1.416^{***} \end{array}$	(3.902) -0.191** (-2.160) 0.069** (2.137) 1.156*** (6.446) 0.090*** (5.388) 0.005
UAI SPITEM×UAI ROA SIZE MBV LEV GROWTH EXPOSURE	0.235** (2.168) 0.0321** (3.302) 0.650*** (9.755) 0.069*** (12.219) 0.006*** (4.878) 1.469*** (41.338)	$\begin{array}{c} 0.253^{**} \\ (2.177) \\ -0.364 \\ (-0.329) \\ 0.568^{***} \\ (7.881) \\ 0.067^{***} \\ (11.021) \\ 0.006^{***} \\ (4.659) \\ 1.416^{***} \end{array}$	-0.191** (-2.160) 0.069** (2.137) 1.156*** (6.446) 0.090*** (5.388) 0.005
SPITEM×UAI ROA SIZE MBV LEV GROWTH EXPOSURE	(2.168) 0.0321** (3.302) 0.650*** (9.755) 0.069*** (12.219) 0.006*** (4.878) 1.469*** (41.338) 1.401***	$\begin{array}{c} (2.177) \\ -0.364 \\ (-0.329) \\ 0.568^{***} \\ (7.881) \\ 0.067^{***} \\ (11.021) \\ 0.006^{***} \\ (4.659) \\ 1.416^{***} \end{array}$	(-2.160) 0.069** (2.137) 1.156*** (6.446) 0.090*** (5.388) 0.005
SPITEM×UAI ROA SIZE MBV LEV GROWTH EXPOSURE	0.0321** (3.302) 0.650*** (9.755) 0.069*** (12.219) 0.006*** (4.878) 1.469*** (41.338)	-0.364 (-0.329) 0.568*** (7.881) 0.067*** (11.021) 0.006*** (4.659) 1.416***	0.069** (2.137) 1.156*** (6.446) 0.090*** (5.388) 0.005
ROA SIZE MBV LEV GROWTH EXPOSURE	(3.302) 0.650*** (9.755) 0.069*** (12.219) 0.006*** (4.878) 1.469*** (41.338)	(-0.329) 0.568*** (7.881) 0.067*** (11.021) 0.006*** (4.659) 1.416***	(2.137) 1.156*** (6.446) 0.090*** (5.388) 0.005
ROA SIZE MBV LEV GROWTH EXPOSURE	0.650*** (9.755) 0.069*** (12.219) 0.006*** (4.878) 1.469*** (41.338)	0.568*** (7.881) 0.067*** (11.021) 0.006*** (4.659) 1.416***	1.156*** (6.446) 0.090*** (5.388) 0.005
SIZE MBV LEV GROWTH EXPOSURE	(9.755) 0.069*** (12.219) 0.006*** (4.878) 1.469*** (41.338)	(7.881) $0.067^{***}$ (11.021) $0.006^{***}$ (4.659) $1.416^{***}$	(6.446) 0.090*** (5.388) 0.005
SIZE MBV LEV GROWTH EXPOSURE	0.069*** (12.219) 0.006*** (4.878) 1.469*** (41.338)	0.067*** (11.021) 0.006*** (4.659) 1.416***	0.090*** (5.388) 0.005
MBV LEV GROWTH EXPOSURE	(12.219) 0.006*** (4.878) 1.469*** (41.338)	(11.021) 0.006*** (4.659) 1.416***	(5.388) 0.005
MBV LEV GROWTH EXPOSURE	0.006*** (4.878) 1.469*** (41.338)	0.006*** (4.659) 1.416***	0.005
LEV GROWTH EXPOSURE	(4.878) 1.469*** (41.338)	(4.659) 1 416***	
LEV GROWTH EXPOSURE	1.469*** (41.338)	1 416***	(1.497)
GROWTH EXPOSURE	(41.338)	1.110	1.703***
GROWTH EXPOSURE	1 101***	(35.773)	(20.704)
EXPOSURE	1.401***	1.443***	1.189***
EXPOSURE	(73.383)	(68.453)	(25.842)
Constant	-0.625***	-0.645***	-0.447***
Constant	(-21,063)	(-19 950)	(-5.830)
Constant	-3 577***	-3.567***	-3.377***
	(-23 810)	(-22, 548)	(-5 631)
		(22.010)	( 5.051)
Observations	41,461	34.883	6.419
R-squared	0.169	0.169	0.172
Number of Firms	7.014	5.906	1.089
Industry FE	YES	YES	YES
Year FE	YES	YES	YES
Country FE	YES	YES	YES
Note: t-statistics in parenthes	es. *** p<0.01. ** p<0.0	5. * p<0.1	125
	52		

# Table 6. Classification shifting and interaction between SPITEM and uncertainty

Page 53 of 60

(1) EU28 (2) WEU (3) EEU VARIABLES UE CE UE CE UE CE SPITEM 0.877*** 0.903* 0.174*** (3.154) (1.665) (3.251) MASC -0.595*** -0.841*** -0.103*** (-3.892) (-4.505) (-3.374) SPITEM×MASC -0.088* -0.851 -0.053*** (-1.759) (-0.589) (-3.644) ROA 0.649*** 0.574*** 1.156*** (-1.759) (-0.589) (-3.644) SIZE 0.074*** 0.073*** 0.091*** (-1.759) (-0.7972) (-0.440) SIZE 0.074*** 0.073*** 0.091*** (-1.779) (11.737) (-5.451) MBV 0.006*** 0.006*** 0.0005 (-1.486) (-1.486) (-1.464) (-1.779) (-1.737) (-5.451) MBV 0.006*** 1.418*** 1.707*** (-1.991***) (-1.864) (-1.418*** (-1.671***) (-1.469)*** (-1.4137*** 1.186*** (-2.091*) (-1.9.807) (-5.854) (-2.0949) (-1.9.807) (-5.854) (-2.0949) (-1.9.807) (-5.854) (-2.0949) (-1.9.807) (-5.854) (-2.0949) (-1.9.807) (-5.854) (-2.0949) (-1.9.168) (-17.711) Observations 41,461 34,883 (-1.419 (-1.771)) Observat	VARLABLES         UE CE         UE CE         UE CE         UE CE           SPITEM         (3,154)         (1,665)         (3,251)           MASC         (-0,59***)         (0,841****)         (0,103****)           SPITEM×MASC         (-0,87)***         (0,841***)         (0,103****)           SPITEM×MASC         (-0,759)         (-6,589)         (-3,644)           ROA         (-649***)         (-579***)         (-6,464)           SIZE         (0,074***)         (0,073***)         (0,091***)           NBV         (0,06***)         (0,006***)         (0,005)           SIZE         (0,074***)         (0,073***)         (0,091***)           MBV         (0,06***)         (0,006***)         (0,005)           SIZE         (-0,770)         (1,1737)         (-5,854)           GROWTH         1397***         (-1,468***)         (-1,771)           Obscrvations         41,461         34,883         6,419           Number of Firms         7,014         5,906         1,089           Industry FE         YES         YES         YES           YES         YES         YES         YES           YES         YES         YES         YES				
VARIABLES         UE CE	VARIABLES         UF, CE         UF, CE         UF, CE         UF, CE           SPITEM         0.877***         0.903*         0.174***           MASC         0.595***         0.841***         0.103***           MASC         0.685*         0.851         -0.053***           SPITEM         MASC         0.088*         -0.851         -0.053***           SPITEM         MASC         0.088*         -0.851         -0.053***           ROA         0.449***         0.574***         1.155***         (1.55***           ROA         0.449***         0.7972)         (6.440)           SIZE         0.074***         0.0073***         0.091***           MBV         0.006***         0.006***         0.005           MBV         0.006***         0.005         (1.779)         (1.137)           MBV         0.006***         0.005         (2.583)         (2.0761)           GROWTH         1.397***         1.487***         1.707***         (2.582)           Constant         -2.987***         -2.772***         -3.468***           Constant         -2.987***         -2.772***         -3.468***           Constant         7.25.578         C19.168 <td< td=""><td></td><td>(1) EU28</td><td>(2) WEU</td><td>(3) EEU</td></td<>		(1) EU28	(2) WEU	(3) EEU
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	SPITEM       0.877***       0.903*       0.174***         MASC       0.595***       0.841***       0.013***         SPITEM×MASC       0.088*       0.851       0.063***         ROA       0.649***       0.574***       1.156***         ROA       0.649***       0.574***       1.156***         ROA       0.649***       0.574***       1.156***         ROA       0.649***       0.574***       1.156***         ROA       0.649***       0.6750       (7.972)         SIZE       0.074***       0.073***       0.091***         MBV       0.006***       0.006***       0.005         MBV       0.006***       0.006***       0.005         GROWTH       1.397***       1.437***       1.186***         Constant       (-2.578)       (-19.168)       (-17.71)         Observations       41.461       34.883       6.419         R-squarcd       0.170       0.170       0.172         Number of Firms       7.014       5.906       1.089         Idustry FE       YES       YES       YES         YES       YES       YES       YES         Car FE       YES       YESY	VARIABLES	UE_CE	UE_CE	UE_CE
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	MASC       0.595***       0.841***       0.103***         SPITEM&MASC       0.088*       0.851       0.003***         ROA       0.649***       0.574***       1.156***         ROA       0.649***       0.574***       1.156***         SIZE       0.73*0       (7.972)       (6.440)         SIZE       0.74***       0.006***       0.000         MBV       0.006***       0.006***       0.005         LEV       (14.779)       (11.377)       (5.451)         MBV       0.006***       0.006***       0.005         LEV       (14.69***       1.418***       1.707***         GROWTH       (13.979)       (68.121)       (25.823)         EXPOSURE       0.622***       -0.640***       -0.439***         Constant       -2.987***       -2.772***       -3.468***         Constant       -2.987***       -2.772***       -3.468***         Constant       -2.987***       -2.772***       -3.468***         Constant       -2.987***       -2.772***       -3.468***         Constant       -2.987***       -2.772***       -3.68***         Constant       -2.987***       -19.168       (17.71)      N	SPITEM	0.877***	0.903*	0.174***
$\begin{array}{llllllllllllllllllllllllllllllllllll$	MASC       -0.595***       -0.841***       -0.103***         SPITEMAMASC       -0.088*       -0.851       -0.053***         ROA       -0.649***       0.574***       1.156***         ROA       -0.649***       0.574***       1.156***         SIZE       -0.7500       (7.972)       (6.440)         SIZE       -0.771**       0.006***       0.005         MBV       -0.006***       0.006***       0.005         LEV       -1.489**       1.437***       1.186***         GROWTH       1.397***       1.437***       1.186***         Constant       -2.987***       -2.772***       -3.468***         Constant       -2.987***       YES       YES<		(3.154)	(1.665)	(3.251)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Grading       (-3,892)       (-4,505)       (-3,374)         SPTEM×MASC       -0,088*       -0,851       -0,053**         ROA       0,649***       0,574***       1,156***         SIZE       0,0750)       (-9,72)       (6,440)         SIZE       0,074***       0,073**       0,091***         MBV       0,006***       0,006***       0,006         LEV       1,469***       1,418***       1,707***         GROWTH       (3,3070)       (6,8121)       (2,823)         EXPOSURE       -(20,949)       (-19,807)       (-5,854)         Constant       2,987***       -2,772***       -3,468***         Constant       2,987***       -2,772***       -3,468***         Constant       2,987***       -2,772***       -3,468***         Constant       2,987***       -2,72***       -3,468***         Constant       2,987***       -2,72***       -3,468***         Constant       2,987***       -2,72***       -3,468***         Constant       2,987***       -2,72***       -3,468***         Constant       2,987**       -1,168)       (-17,711)         Observations       1,140       5,906       1,989	MASC	-0.595***	-0.841***	-0.103***
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	SPITEM×MASC       -0.088*       -0.851       -0.053***         ROA       0.649***       0.574***       1.156***         ROA       0.649***       0.071***       0.091***         SIZE       0.074***       0.073***       0.091***         MBV       0.066***       0.006***       0.005         MBV       0.066***       0.006***       0.005         LEV       (1.779)       (1.137)       (5.451)         MBV       0.066***       0.006***       0.005         GROWTH       (1.397***       1.418***       1.707***         GROWTH       (13.070)       (6.121)       (25.823)         EXPOSURE       -0.622***       -0.640***       -0.49***         (-20.949)       (19.807)       (-5.854)       (-7.711)         Observations       41.461       34.883       6.419         R-squared       0.170       0.172       Number of Firms       7.014       5.906       1.089         Industry FE       YES       Y		(-3.892)	(-4.505)	(-3.374)
ROA $(-1.759)$ $(-0.589)$ $(-3.644)$ ROA $0.649^{***}$ $0.574^{***}$ $1.156^{***}$ SIZE $0.073^{***}$ $0.091^{***}$ MBV $0.06^{***}$ $0.006^{***}$ 0.005 $(4.886)$ $(4.648)$ (12.779) $(11.737)$ $(5.451)$ MBV $0.006^{***}$ $0.006^{***}$ 0.005 $(4.886)$ $(4.648)$ (14.91) $(1.491)$ LEV $1.469^{***}$ $1.418^{***}$ $1.707^{***}$ $(41.401)$ $(35.863)$ $(20.761)$ GROWTH $1.397^{***}$ $1.437^{***}$ $1.86^{***}$ $(73.070)$ $(68.121)$ $(25.823)$ EXPOSURE $-0.622^{***}$ $-0.640^{***}$ $-0.499^{***}$ $(-20.949)$ $(-19.807)$ $(-5.854)$ Constant $-2.987^{***}$ $-2.987^{***}$ $-2.772^{***}$ $-3.468^{***}$ $(-25.578)$ $(-19.168)$ $(-17.711)$ Observations $41.461$ $34.883$ $6.419$ R-squared $0.170$ $0.170$ $0.172$ Number of Firms $7.014$ $5.906$ $1.089$ Industry FEYESYESYESYear FEYESYESYESNote: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	ROA       0.649***       0.574***       1.156***         SIZE       0.074***       0.073***       0.091***         MBV       0.006***       0.006***       0.005         MBV       0.006***       0.006***       0.005         LEV       1.469***       1.418***       1.707***         MBV       0.006***       0.006***       0.005         GROWTH       1.397***       1.437***       1.186***         CADSURE       (-20.949)       (-19.807)       (-5.854)         CONStant       (-2.987***       -2.772***       -3.468***         (-25.578)       (-19.168)       (-17.71)         Observations       41.461       34.883       6.419         Number of Firms       7.014       5.906       1.089         Industry FE       YES       YES       YES         Year FE       YES       YES       YES         Outry FE       YES       YES       YES	SPITEM×MASC	-0.088*	-0.851	-0.053***
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	ROA         0.649***         0.574***         1.156***           SIZE         0.0730)         (7.972)         (6.440)           SIZE         0.074***         0.005           MBV         0.006***         0.006***         0.005           MBV         0.006***         0.006***         0.005           LEV         1.469***         1.418***         1.707***           LEV         1.469***         1.418***         1.707***           GROWTH         1.397***         1.438***         1.707***           GROWTH         1.397***         1.438***         1.707***           Constant         -0.622***         -0.640***         -0.449***           Constant         -2.987***         -2.772***         -3.468***           Constant         -2.987***         -2.772***         -3.468***           Constant         -2.987***         -2.772***         -3.468***           Muburp of Firms         7.014         5.906         1.089           Industry FE         YES         YES         YES           YES         YES         YES         YES           YES         YES         YES           Country FE         YES         YES		(-1.759)	(-0.589)	(-3.644)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	International (0,750)       (7,972)       (6,440)         SIZE       0.074***       0.073***       0.091***         MBV       0.006***       0.006***       0.005         MBV       0.006***       0.006***       0.005         LEV       1.469***       1.418***       1.707***         JEV       1.469***       1.437***       1.186***         GROWTH       1.397***       1.437***       1.186***         EXPOSURE       -0.622***       -0.640***       -0.449***         Constant       -2.987***       -2.772***       -3.468***         Constant       -2.987***       -2.772***       -3.468***         Number of Firms       7.014       34.883       6.419         Number of Firms       7.014       5.906       1.039         Idatity FE       YES       YES       YES       YES         Year FE       YES       YES       YES       YES         Net: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	ROA	0 649***	0.574***	1 156***
SIZE $0.074^{***}$ $0.073^{***}$ $0.091^{***}$ MBV $0.006^{***}$ $0.006^{***}$ $0.005$ (4.886) (4.648) (1.491) LEV $1.469^{***}$ $1.418^{***}$ $1.707^{***}$ (41.401) (35.863) (20.761) GROWTH $1.397^{***}$ $1.437^{***}$ $1.186^{***}$ (73.070) (68.121) (25.823) EXPOSURE $-0.622^{***}$ $-0.640^{***}$ $-0.449^{***}$ (-20.949) (-19.807) (-5.854) Constant $-2.987^{***}$ $-2.772^{***}$ $-3.468^{***}$ (-25.578) (-19.168) (-17.711) Observations $41,461$ $34,883$ $6,419$ R-squared $0.170$ $0.170$ $0.172$ Number of Firms $7,014$ $5,906$ $1,089$ Industry FE YES YES YES Year FE YES YES YES YES Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	SIZE         0.074***         0.073***         0.001***           MBV         (12,779)         (11,737)         (5,451)           MBV         (4.886)         (4.648)         (1.491)           LEV         1.469***         1.418***         1.707***           GROWTH         1.397***         1.437***         1.136***           GROWTH         (73.070)         (68.121)         (25.823)           EXPOSURE         -0.622***         -0.640***         -0.449***           (-20.949)         (-19.168)         (-17.11)           Observations         41.461         34.883         6.419           R-squared         0.170         0.170         0.172           Number of Firms         7.014         5.906         1.089           Idustry FE         YES         YES         YES           Outry FE         YES         YES         YES		(9 750)	(7 972)	(6 440)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Date       (12,779)       (11,77)       (5,451)         MBV       0.006***       0.005         (4,886)       (4,648)       (1491)         LEV       (4,69***)       1.418***         LEV       (4,69***)       1.418***         GROWTH       1.397***       1.437***         1.307***       1.437***       1.186***         GROWTH       (20,949)       (68.121)       (25.823)         EXPOSURE       -0.622***       -0.640***       -0.449***         (-20,949)       (-19.807)       (-5.854)         Constant       -2.987***       -2.772***       -3.468**         (-25.578)       (-19.168)       (-17.711)         Observations       41,461       34,883       6,419         R-squared       0.170       0.172       0.170       0.172         Number of Firms       7,014       5,906       1,089       1         Idustry FE       YES       YES       YES       YES         Country FE       YES       YES       YES       YES         Oute: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	SIZE	0 074***	0 073***	0.091***
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	MBV         0.006***         0.006***         0.005           LEV         1.469***         1.418***         1.707***           JLEV         1.469***         1.418***         1.707***           GROWTH         1.397***         1.437***         1.186***           CAUSTRE         -0.622***         -0.640***         -0.449***           CODSURE         -0.622***         -2.772***         -3.468***           CODSURE         -0.622***         -2.772***         -3.468***           CODSURE         -1.2987***         -2.772***         -3.468***           CODSURT         CODSURT         0.170         0.172           Number of Firms         7.014         5.906         1.089           Jduatry FE         YES         YES         YES           Vers t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1		(12 779)	(11737)	(5451)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Intro       0.000       0.000       0.000         LEV       1.469**       1.418**       1.707***         GROWTH       1.397***       1.438**       1.707***         GROWTH       (73.070)       (68.121)       (25.823)         EXPOSURE       -0.622***       -0.640***       -0.449***         Constant       (-20.949)       (-19.807)       (-5.854)         Constant       (-25.578)       (-19.168)       (-17.711)         Observations       41.461       34.883       6.419         R-squared       0.170       0.170       0.172         Number of Firms       7.014       5.906       1.089         Industry FE       YES       YES       YES         Year FE       YES       YES       YES         Outry FE       YES       YES       YES         Vet: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	MBV	0.006***	0.006***	0.005
LEV $(1.469^{**} 1.418^{**} 1.707^{**} (41.401)$ $(35.863)$ $(20.761)$ GROWTH $1.397^{**} 1.437^{**} 1.186^{***}$ (73.070) $(68.121)$ $(25.823)EXPOSURE -0.622^{***} -0.640^{***} -0.449^{***}(-20.949)$ $(-19.807)$ $(-5.854)Constant -2.987^{***} -2.772^{***} -3.468^{***}(-25.578)$ $(-19.168)$ $(-17.711)Observations 41,461 34,883 6,419R-squared 0.170 0.170 0.172Number of Firms 7,014 5,906 1,089Industry FE YES YES YESYear FE YES YES YES YESYear FE YES YES YES YESNote: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1$	LEV       (469***       (418***       107***         GROWTH       (1397***       (148***)       118****         GROWTH       (1397***       (148***)       (107***)         EXPOSURE       (052***       -0.640***       -0.449****         Constant       (-20949)       (-19.807)       (-5.854)         Constant       (-25.578)       (-10.168)       (-17.711)         Observations       41,461       34,883       6,419         R-squared       0.170       0.170       0.172         Number of Firms       7.014       5.906       1.089         Industry FE       YES       YES       YES         YES       YES       YES       YES         Country FE       YES       YES       YES         Outry FE       YES       YES       YES         Vit: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1		(4 886)	(4.648)	(1.491)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	LL (14) (41,401) (43,863) (20,761) GROWTH 1397** 1437*** 1,186*** (73,070) (68,121) (25,823) EXPOSURE -0.622*** -0.640*** -0.449*** (20,949) (-19,807) (-5,854) Constant -2.987*** -2.772*** -3.468*** (-25,578) (-19,168) (-17,711) Observations 41,461 34,883 6,419 R-squared 0,170 0,170 0,172 Number of Firms 7,014 5,906 1,089 Industry FE YES YES YES Yes YES YES YES Country FE YES YES YES Country FE YES YES YES Country FE YES YES YES Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	I FV	1 469***	1 418***	1 707***
GROWTH $1.397^{***}$ $1.437^{***}$ $1.186^{***}$ $(73.070)$ $(68.121)$ $(25.823)$ EXPOSURE $-0.622^{***}$ $-0.640^{***}$ $-0.449^{***}$ $(-20.949)$ $(-19.807)$ $(-5.854)$ Constant $-2.987^{***}$ $-2.772^{***}$ $-3.468^{***}$ $(-25.578)$ $(-19.168)$ $(-17.711)$ Observations $41.461$ $34,883$ $6,419$ R-squared $0.170$ $0.170$ $0.172$ Number of Firms $7,014$ $5,906$ $1,089$ Industry FEYESYESYESYear FEYESYESYESOutry FEYESYESYESNote: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	GROWTH       1,37***       1,437***       1,186***         GROWTH       (73,070)       (68,121)       (25,823)         EXPOSURE       -0,622***       -0,640***       -0,449***         (-20,949)       (-19,807)       (-5,854)         Constant       -2,987***       -2,772***       -3,468***         (-25,578)       (-19,168)       (-17,711)         Observations       41,461       34,883       6,419         R-squared       0.170       0.172       Number of Firms       7,014       5,906       1,089         Industry FE       YES       YES       YES       YES       YES       YES       YES         Year FE       YES       YES <td< td=""><td></td><td>(1, 40)</td><td>(35.863)</td><td>(20.761)</td></td<>		(1, 40)	(35.863)	(20.761)
OROWTH1.397421.4374241.18644(73.070)(68.121)(25.823)EXPOSURE $-0.622^{***}$ $-0.640^{***}$ (-20.949)(-19.807)(-5.854)Constant $-2.987^{***}$ $-2.772^{***}$ $-3.468^{***}$ (-25.578)(-19.168)Observations41,46134,883R-squared0.1700.172Number of Firms7,0145,906Industry FEYESYESYear FEYESYESYear FEYESYESNote: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	GKOW III       (137) 070)       (143) (21)       (125) (25) (25) (25) (25) (25) (25) (25) (	CDOWTH	(41.401)	(55.805)	(20.701) 1 196***
EXPOSURE $-0.622^{***}$ $-0.640^{***}$ $-0.449^{***}$ (-20.949) $(-19.807)$ $(-5.854)Constant -2.987^{***} -2.772^{***} -3.468^{***}(-25.578)$ $(-19.168)$ $(-17.711)Observations 41,461 34,883 6,419R-squared 0.170 0.170 0.172Number of Firms 7,014 5,906 1,089Industry FE YES YES YESYear FE YES YES YESYear FE YES YES YESNote: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1$	EXPOSURE       (-0.622**)       -0.640***       (-2.0.2.3)         Constant       -2.987***       -2.772***       -3.468***         (-25.578)       (-19.168)       (-17.711)         Observations       41.461       34.883       6.419         R-squared       0.170       0.172       Number of Firms         Number of Firms       7.014       5.906       1.089         Industry FE       YES       YES       YES         Year FE       YES       YES       YES         Outry FE       YES       YES       YES         Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	UKUWIII	(72,070)	(69, 121)	(25, 922)
EXPOSORE -0.622*** -0.040*** -0.449*** (-20.949) (-19.807) (-5.854) Constant -2.987*** -2.772*** -3.468*** (-25.578) (-19.168) (-17.711) Observations 41,461 34,883 6,419 R-squared 0.170 0.170 0.172 Number of Firms 7,014 5,906 1,089 Industry FE YES YES YES Year FE YES YES YES YES Year FE YES YES YES YES Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	EXPOSURE       -0.022***       -0.040***       -0.449***         (-20.949)       (-19.807)       -0.5854)         Constant       -2.987***       -2.772***       -3.468***         (-25.578)       (-19.168)       (-17.711)         Observations       41,461       34,883       6.419         R-squared       0.170       0.172         Number of Firms       7.014       5.906       1.089         Industry FE       YES       YES       YES         Year FE       YES       YES       YES         Country FE       YES       YES       YES         Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	EVDOLIDE	(73.070)	(00.121)	(23.823)
Constant $\begin{array}{c} (-20.949) \\ -2.987^{***} \\ (-25.578) \\ (-19.168) \\ (-17.711) \end{array}$ Observations $\begin{array}{c} 41,461 \\ -25.578) \\ (-19.168) \\ (-17.711) \\ (-5.854) \\ (-17.711) \\ (-$	Constant       -2.987**       -2.772***       -3.468***         (-25.578)       (-19.168)       (-17.711)         Observations       41,461       34,883       6,419         R-squared       0.170       0.170       0.172         Number of Firms       7,014       5,906       1,089         Industry FE       YES       YES       YES         Year FE       YES       YES       YES         Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	EXPOSURE	-0.622	$-0.040^{***}$	-0.449***
Constant $-2.98/***$ $-2.7/2***$ $-3.468***$ (-25.578)(-19.168)(-17.711)Observations41,46134,8836,419R-squared0.1700.1700.172Number of Firms7,0145,9061,089Industry FEYESYESYESYear FEYESYESYESCountry FEYESYESYESNote: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	Constant         -2.98/***         -2.1/2***         -3.468***           (-25.578)         (-19.168)         (-17.711)           Observations         41,461         34,883         6,419           R-squared         0.170         0.170         0.172           Number of Firms         7,014         5,906         1,089           Industry FE         YES         YES         YES           Year FE         YES         YES         YES           Country FE         YES         YES         YES           Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	0	(-20.949)	(-19.80/)	(-5.854)
(-25.578)       (-19.168)       (-17.711)         Observations       41,461       34,883       6,419         R-squared       0.170       0.170       0.172         Number of Firms       7,014       5,906       1,089         Industry FE       YES       YES       YES         Year FE       YES       YES       YES         Country FE       YES       YES       YES         Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	(-12,578)       (-19,168)       (-17,711)         Observations       41,461       34,883       6,419         R-squared       0.170       0.172       0.170       0.172         Number of Firms       7,014       5,906       1,089         Industry FE       YES       YES       YES         Year FE       YES       YES       YES         Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	Constant	-2.98/***	-2.//2***	-3.468***
Observations $41,461$ $34,883$ $6,419$ R-squared $0.170$ $0.172$ Number of Firms $7,014$ $5,906$ $1,089$ Industry FE       YES       YES       YES         Year FE       YES       YES       YES         Country FE       YES       YES       YES         Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	Observations 41,461 34,883 6,419 R-squared 0.170 0.172 Number of Firms 7,014 5,906 1,089 Industry FE YES YES YES Year FE YES YES YES Country FE YES YES YES Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1		(-25.578)	(-19.168)	(-17.711)
R-squared0.1700.172Number of Firms7,0145,906Industry FEYESYESYear FEYESYESCountry FEYESYESVote: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	R-squared 0.70 0.72 Number of Firms 7,014 5,906 1,089 Industry FE YES YES YES Year FE YES YES YES Country FE YES YES YES Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	Observations	41 461	34 883	6 4 1 9
Number of Firms7,0145,9061,089Industry FEYESYESYESYear FEYESYESYESCountry FEYESYESYESNote: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	Number of Firms 7,014 5,906 1,089 Industry FE YES YES YES YES YES YES YES YES YES YE	R-squared	0.170	0 170	0,172
Industry FE YES YES YES YES YES YES YES YES YES YE	Industry FE YES YES YES Year FE YES YES YES YES Country FE YES YES YES Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	Number of Firms	7.014	5 906	1 080
Year FE YES YES YES YES Country FE YES YES YES YES Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	Year FE YES YES YES YES Country FE YES YES YES Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	Industry FE	VES	5,900 VES	VES
YES     YES     YES       Country FE     YES     YES       Vote: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	fear FE TES TES TES TES Country FE YES YES Vote: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	Industry FE	I ES VES	I ES VES	I ES VES
Country FE FES FES FES FES FES VIES	Country FE <u>1ES 1ES 1ES</u> Note: t-statistics in parentheses, *** p<0.01, ** p<0.05, * p<0.1	Year FE	I ES VES	I ES	I ES VES
	53				
			53		

#### Table 7. Classification shifting and interaction between SPITEM and masculinity.

	(1) EU28	(2) WEU	(3) EEU	
VARIABLES	UE CE	UE CE	UE CE	
SPITEM	0.773***	0.953*	0.207**	
	(2.813)	(1,740)	(2.079)	
IND	0 101**	0 328**	-0.122***	
	(2.094)	(3.860)	(-2, 929)	
SDITEMAIND	(2.0)4)	(5.800)	0.0021***	
SFILEMAIND	(1, 727)	(0.676)	(2,650)	
DOA	(1./2/)	(0.070)	(3.039)	
ROA	0.651***	0.368***	$1.1/1^{***}$	
	(9.781)	(/.8/4)	(6.537)	
SIZE	0.069***	0.066***	0.093***	
	(12.114)	(10.871)	(5.613)	
MBV	0.006***	0.006***	0.005	
	(4.895)	(4.680)	(1.433)	
LEV	1.469***	1.416***	1.712***	
	(41.392)	(35.819)	(20.786)	
GROWTH	1.403***	1.445***	1.199***	
	(73.576)	(68,583)	(26.169)	
EXPOSURE	-0 626***	-0 645***	-0.451***	
	(-21, 0.89)	(-19 965)	(-5.876)	
Constant	(21.00))	2 / 27***	2 81/***	
Constant	(56.271)	(54.146)	(10.468)	
	(-30.371)	(-34.140)	(-10.408)	
Observations	41,461	34,883	6,419	
R-squared	0.169	0.169	0.174	
Number of Firms	7 014	5 906	1 089	
Industry FE	YES	VES	YES	
Vear FE	VES	VES	VFS	
Country FE	VES	VES		
lote: t-statistics in parenthe	$\frac{120}{888} \times 1001 \times 1001$	$\frac{110}{05 * n < 0.1}$	125	
I				
	54			
	0.			

<b>Fable 8. Classi</b>	fication shifting	and interaction	between	<b>SPITEM and</b>	individualism.
------------------------	-------------------	-----------------	---------	-------------------	----------------

Page 55 of 60

	(1) EU28	(2) WEU	(3) EEU	
VARIABLES	UE CE	UE CE	UE CE	
SPITEM	0 329**	0.204***	0 133**	
	(2.164)	(3 479)	(2.080)	
PD	-0 178***	-0 244***	-0 149***	
	$(-4\ 409)$	(-4 679)	(-2,739)	
SPITEM×PD	-0.0370**	0 0737**	0 0297**	
SITTEMINI D	(-2,138)	(2.085)	(2 0 2 9)	
ROA	0.912***	0.842***	1 159***	
Rom	(10.564)	(8 464)	(6 4 6 6)	
SIZE	0.070***	0.067***	0.089***	
SILL	(10,779)	(9.371)	(5,359)	
MBV	0.007***	0.007***	0.005	
	(4,704)	(4505)	(1.498)	
IFV	1 522***	1 443***	1 704***	
LL V	(35,654)	(28.644)	(20,733)	
GROWTH	1.207***	1 265***	1 101***	
	(57 776)	(51.004)	(26.026)	
EVDOCUDE	(57.770)	0.652***	(20.020)	
EAFOSURE	-0.010	(16.270)	(5, 912)	
Constant	(-17.400)	(-10.379) 2.177***	(-3.013)	
Constant	$-2.303^{+14}$	$-2.1//^{-100}$	$-3.239^{++++}$	
	(-11.914)	(-8.427)	(-/.901)	
Observations	28 625	22 047	6 410	
Duser various P squared	28,025	22,047	0,419	
Number of Firms	0.109	0.108	1.020	
Industry EE	5,200 VES	4,160 VES	1,069 VES	
HIGUSUY FE	I ES VES	I ES VES	I ES VES	
Year FE	Y ES VES	YES	YES VES	
<u>Country TE</u>	1 LS	* n<0.1	ILS	
	55			

#### Table 9. Classification shifting and interaction between SPITEM and power distance.

2	
2	
3	
5	
4	
5	
5	
6	
7	
/	
8	
0	
9	
10	
11	
11	
12	
1 2	
13	
14	
15	
16	
10	
17	
18	
10	
19	
20	
20	
21	
<u>ວ</u> ว	
۲۲	
23	
24	
24	
25	
26	
20	
27	
20	
20	
29	
20	
30	
31	
22	
32	
33	
24	
34	
35	
22	
36	
37	
57	
38	
30	
59	
40	
41	
11	
42	
43	
44	
15	
75	
46	
Δ7	
48	
<u>4</u> 0	
50	
51	
51	
52	
53	
55	
54	
55	
55	
56	
57	
57	

Table 10.	<b>Classification</b>	shifting and i	nteraction b	etween SPI	TEM and lo	ng-term
orientatio	on.					

	(1) EU28	(2) WEU	(3) EEU	
VARIABLES	UE CE	UE CE	UE CE	
SPITEM	0.502**	0.136**	0 542***	
STILM	(3.510)	(2 121)	(2732)	
ITO	0.462***	(2.121) 0 $(1.127)$	(2.752)	
	(5, 221)	(4.744)	(1.200)	
CDITENA I TO	(3.221) 0.502***	(4./44)	(1.099)	
SPITEM×LTO	0.503	0.830**	-0.101***	
	(2.865)	(2.191)	(-2.001)	
ROA	1.153***	1.142***	1.174***	
	(14.334)	(12.581)	(6.543)	
SIZE	0.083***	0.082***	0.090***	
	(13.634)	(12.298)	(5.494)	
MBV	0.007***	0.007***	0.005	
	(5.315)	(5.220)	(1.470)	
LEV	1.640***	1.610***	1.712***	
	(41.223)	(35,188)	(20.805)	
GROWTH	1 343***	1 382***	1 196***	
	(64 291)	(58 520)	(26.031)	
EXPOSURE	-0 684***	_0 727***	-0 456***	
EAIOSOKE	(20.054)	(20.028)	-0.430	
Constant	(-20.930)	(-20.038)	(-J.930) 5 019***	
Constant	-4.393***	-4.54/***	$-5.018^{+++}$	
	(-27.223)	(-25.847)	(-6.331)	
01	25.072	20.405	6 410	
Observations	35,063	28,485	6,419	
R-squared	0.172	0.172	0.173	
Number of Firms	6,172	5,064	1,089	
Industry FE	YES	YES	YES	
Year FE	YES	YES	YES	
Country FE	YES	YES	YES	
Note: t-statistics in parenthe	ses, *** p<0.01, ** p<0.0	05, * p<0.1		
	56			

Table 11. REM1 and Hofstede - Cultural Variables.

	(1) EU28	(2) WEU	(3) EEU
VARIABLES	REM1	REM1	REM1
JAI	0.179***	0.064	0.332***
	(3.265)	(1.023)	(4.311)
IASC	-0.621***	-0.414***	-0.381***
	(-16.005)	(-7.924)	(-5.938)
ND	0.021***	0.012***	0.337***
	(6.797)	(3.393)	(6.716)
	0 046***	0.037*	-0 425***
	$(4\ 040)$	(1.725)	(-4 966)
ТО	0.042*	0 104***	5 513***
	(1.756)	(2.915)	(6.818)
	0.840***	0.849***	0 784***
	(57,406)	(51,022)	(24.032)
76	(37.400)	(31.922) 0.004***	(24.932)
	(2.045)	(2.578)	(0.283)
	(2.043)	(3.378)	(0.283)
DV	(1.775)	(0.402)	$0.001^{\circ}$
717	(1.//3) 0.1(2***	(U.4U3)	(1.824)
L V	$0.163^{**}$	$0.225^{***}$	0.052***
DOUTU	(19.217)	(22.435)	(3.126)
ROWTH	0.4/1***	0.456***	0.48//***
	(101.908)	(87.710)	(50.047)
XPOSURE	-0.129***	-0.152***	-0.082***
	(-17.978)	(-19.519)	(-4.797)
onstant	0.099	-0.016	-9.699***
	(1.107)	(-0.099)	(-6.384)
oservations	42.411	31.558	10.578
-squared	0 350	0 351	0.352
imber of firms	4 476	3 235	1 221
dustry FF	VFS	VFS	VFS
ar FE	YES	YES	YES
untry FF	VES	VES	VES
e: t-statistics in parenthes	rES es, *** p<0.01, ** p<0.0	05, * p<0.1	1125
	57		

Table 12.	<b>REM2</b> and	Hofstede -	Cultural '	Variables.

VADIADI DO	(1) EU28	(2) WEU	(3) EEU
VARIABLES	REM2	REM2	REM2
UAI	-0.023	-0.025***	-2.042***
	(-1.421)	(-3.521)	(-8.016)
MASC	0.176***	0.114***	-0.224***
	(15.563)	(8.397)	(-2.797)
IND	-0.005***	-0.003***	-0.114***
	(-5.646)	(-2.692)	(-7.677)
PD	-0.031***	-0.029***	0.208***
	(-9.456)	(-5.367)	(8.008)
LTO	-0.019***	-0.035***	-0.420***
	(-2.700)	(-3.911)	(-5.879)
ROA	-0.070***	-0.056***	-0.095***
	(-16.711)	(-13.490)	(-8.650)
SIZE	0.014***	0.010***	0.043***
	(42.402)	(33.211)	(31.751)
MBV	-0.000	-0.000	-0.000
	(-0.555)	(-0.422)	(-0.214)
LEV	-0.037***	-0.047***	-0.006
	(-14.892)	(-18.314)	(-1.078)
GROWTH	-0.015***	-0.016***	-0.012***
	(-11.293)	(-12.329)	(-3.432)
EXPOSURE	0.038***	0.038***	0.040***
	(17.752)	(18.575)	(6.687)
Constant	-0.298***	-0.212***	2.833***
	(-11.627)	(-5.058)	(6.170)
	()	(1111)	(*****)
Observations	45 836	34 068	11 478
R-squared	0 071	0.065	0 1 3 3
Number of firms	4 574	3 316	1 238
Industry FE	YES	YES	YES
Year FE	YES	YES	YES
Country FE	YES	YES	YES
te <sup>.</sup> t-statistics in parentl	$\frac{120}{120}$	$\frac{110}{5 * n < 0.1}$	1125
ne. t-statistics in parenti	p < 0.01, p < 0.01	<i>J</i> , <i>p</i> < 0.1	
	EO		
	58		

Do cultural differences affect the quality of financial reporting in the EU? An analysis of Western EU countries vis a vis Eastern EU country.

# **Response to Reviewer 2**

We would like to thank the Reviewer for having read and accepted our paper. We have revised the paper inspired by your comments and suggestions.

Our revision is now complete.

We responded to all points in our response.

Please find below our detailed responses to each comment.

Summary

The Reviewer argues that the paper has substantially improved through the revision and the authors have thoroughly addressed all my major concerns and found a satisfactory solution to the issues.

The Reviewer proposes some minor issues that we address below.

Minor concerns

1. Title: The Reviewer suggests using "Eastern" instead of "East" as the authors also use "Western".

Thank you for this comment, we have amended the title accordingly.

2. The authors should use abbreviations only after having introduced them. In particular, the acronym "REM" is first used on page 3 but only a few sentences later, they introduce REM as "real earnings management".

We introduced REM as "real earnings management" two sentences before as suggested by the reviewer. In general, we have checked all abbreviations used in the paper.

3. The footnote no. 1 is irrelevant as they define "classification shifting" only 3 sentences later in the text.

Thank you for this point, we absolutely agree with this, and we removed that footnote.

4. On page 4. The authors state that "This paper [...] proposes to examine whether they explain earnings management in the form of classification shifting and real earnings management". In the abstract, in contrast, they only refer to classification shifting. They need to be more specific: is it classification shifting only or do they also include REM in their analysis?

Thank you for this point. The main analysis is on classification shifting and later in the paper we also verify REM as robustness exercise. So, we slightly modified the point mentioned by the reviewer at page 4 of the paper specifying that the main analysis is on classification shifting and REM is a robustness analysis.

5. The term "EU-new" still appears in the paper (e.g., p. 7)

We changed this point using the new abbreviation (EEU) and in general we controlled the rest of the paper to avoid any other similar typos.

6. Can the authors please review hypothesis H2. Maybe there is a misunderstanding. In 2.2.2 they state that "East European culture is less individualistic than West European (Hofstede, 1980; Kolman et al. 2003)". However, H2 seems to be contradictory to this: "High level of individualism, as associated with EEU countries,...". In my view it should read, "Low level of individualism, as associated with EEU countries,..."

Thank you very much for this point. We redraft hypothesis 2 to address this point and now reads: "Low level of individualism, as associated with EEU countries,..."

7. p. 27: please reconsider the sentence "According to Hofstede et al. (2010), EEU have more men than WEU. "Do the authors mean that there are more men in management positions? This is also the case in WEU, however, maybe it's even more in EEU?

In line with this comment, we delete this reference as it could create a misleading interpretation and it is not part of our identification.

8. Although in their CoE letter they state that they deleted the notion "Authors' estimations" in all tables, it is in fact still to be found under tables 6 to 12.

We removed now "Authors' estimations" also from all tables where it was still included in the previous version of the paper.

9. The manuscript still has some issues with English language and writing, particularly with punctuation and grammar (e.g., p. 2, remove the period before referring to (Akaliyski, 2019); p. 25, "showed" should be changed to "shown", p. 25 the double "in in" should be removed; p. 33, "WEE" should be changed to "WEU"; p. 33 "taken in account" should be changed to "taken into account"; p. 34, again "taken into account" should be used; literature: the source Akaliyski (2019) should apply the same referencing style (remove ":"). I suggest that the authors must fix this before this work can be published.

Thank you for all the above comments that we take into account in the revised manuscript. We also proof read the manuscript for remaining errors.